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学位論文内容の要旨

博士の専攻分野の名称　　博士（理学） 氏名　王　林勝

学　位　論　文　題　名

Studies　on　HZSM－5－Supported　Mo　and　Re　Catalysts一一一Catalytic　Performances

　in　Methane　Aromatization　Reaction　and　Their　Structural　Characterization

　　　　　　　　　　（ゼオライト担持Mo及びRe触媒の研究

　　　一メタンの芳香族化反応の触媒特性と構造キャラクタリゼーション）

　触媒作用を分子レベルで詳細に理解し、より高活性な触媒を分子設計するためには、よ

り良く構造規定された活性中心を固体表面に精密合成するとともに、活性サイト周辺の反

応場をも精密制御することが重要である。金属担持触媒は石油化学原料の酸化反応、異性

化反応及び重合反応などの触媒反応に用いられる工業的に有用な固体触媒であり、その触

媒活性構造や触媒メカニズムに関して基本的な理解を得ることは学術的意義のみならず、

工業的にも重要な研究課題である。天然ガス及びCO・のエネルギー化や資源化は炭素資

源エネルギー体系の改革と関連して21世紀にむけての重要な研究課題である。本研究に

おいて・申請者はゼオライト担持のMo及びRe触媒メタンの脱水i素縮合化反応でベンゼ

ン、ナフタレンの芳香族原料と水素を合成し、高効率で安定な触媒性能を示すことをはじ

めて見出した。新規なゼオライト男持Mo及びRe触媒についてメタンの芳香族化反応の

触媒特性と触媒活性構造のキャラクタリゼーションをXAFS！TEM／XPSなどの物理化学的

手法により明らかにした。また本研究において、Moカ・一・一一・バイド粒子とゼオライトのブレ

ンステット黒点からなる複合触媒作用に関するメタンの脱水素芳香化反応の新しいメカニ

ズムを提出した。

本論文では十章から構成されている。

　第一章では本研究の位置づけを示した後、第二章から三章において、Mo！HZSM－5を触

媒とし、60Cト’800℃の温度域でメタンからベンゼンなどの芳香族化合物と水素を直接合成する

メタンの芳香族化反応について述べる。まず、HZSM・5ゼオライトを担体に用いて20種以上

の異なる遷移金属元素の担持触媒を調整しメタンの芳香族化反応に対する触媒特性を調べた。こ

の結果・Mo・Re・Wといったオレフィンのメ同好シス反応に有効な触媒金属においてのみ、高

いメタン酸化率で、また高い選択率でベンゼンを生成することを見出した。さらに様々なミクロ

及びメソ細孔多孔質担体を用いて調整したMo担持触媒についてメタンの芳香族化反応を調べ

た。その結果・HZSM・5以外の10A以上の細孔径を有するMORやFAUやFSM．16などの
ゼオライトやシリカやアルミナ担体を用いた場合、コーク生成が著しくベンゼンへの転化率が著

しく低いことが判った。最適なSiGJAI・O＝20・70を有するHZSM・5にMoを担持すると特異的

に高活性と高選択率でベンゼン、ナフタレン等の芳香族化合物を生成する事を見いだした。H

ZSM・5上のB酸がベンゼン生成のためのメタンの縮合反応に効果的であると考えた。

　第三章においてMo！HZSM・5とメタンとの反応についてSEM、　TEMおよびXAFS解析に

矯　　　’
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よって調べた。その結果、Mo酸化物はHZSM・5上に均一平坦にに分布しすること、メタンと

反応することで、その粒子がM（hCであることが解った。　EXAFS解析によると生成したM（ta　C

のMo・Mo配位数は2～3であり、　Mo2C結晶でのC，N，＝12に比べて、著しく小さい値をMo2

CはZSM・5ゼオライト細孔チャンネル（6　A径）に高分散されていると考えた。

　第四章では第三章での研究結果をいかして、メタンへのCOあるいはCO，の添加効果と

作用メカニズムについて調べた。

　メタン反応率は反応時間とともに低下するが、COあるいはCO2をメタン中に数％添加

すると、ベンゼンの生成速度の増大とともに、活性低下の度合が著しく改善されことを見出

した。定常状態では、導入したCO2の2倍量のCOが生成していることが分かった。従っ

て、この条件下で　CH，＋CO・ニ2CO＋2H2の反応が進行し、結果的にはCOないしCO、か

ら生成した活性酸素がMoカーバイド表面の不活性なコークと反応し、副生するコークを

低減して、触媒活性の安定化が得られると推察した。

　さらに、第五章では反応後、触媒表面にどのような蓄積炭素種があるのかを空気で昇温し

ながらコーク生成量の解析を行った。その結果、COやCO・添加により触媒上のcoke蓄積

量が著しく減少しており、これが活性低下を防ぐ1つの要因となっていることが判った。

　さらに、第六章においてメタンへのCOやCO2の添加効果がどうして起こるのかを調べ

るために13COをメタンに混合して実験した。その結果、反応進行につれてベンゼン中に
’3

bが均一的に分布しており、ベンゼン1分子中に0．6から0．9個の13Cが含まれていた。

以上の結果から、Mo・C上でCOは2CO＝CO・＋Cの反応により活性な炭素と酸素を供給し、

この活性炭素がメタンの脱水素によって生成した炭素2個を含む中間体を経由してゼオライ

ト表面でベンゼンなどの芳香族化合物へと転化していく。bifunctionalな触媒作用のメカニ

ズムを提案した。

　第七章においてXPSやMo・C＋HZSM－5ハイブリッド触媒の研究からM（hCが本反応

におけるメタンを活性化するサイトであることを見出した。

　第八～九章においては、これまで研究をし

てきたMo∠HZSM・5触媒に対してさらに高活

性で高選択的なメタンの芳香族化反応に有効

なRe！HZSM・5を発見し、この触媒特性と反

応メカニズムについて研究を行った。Re触

媒は、Mo触媒とほぼ同様のメタン転化率や

ベンゼンへの高い生成選択性を示した。さら

に、メタンにCO・を添加すると触媒活性が

安定化し長時間一定の活性を保った。コーク

生成の抑制が触媒の安定化の要因であること

を結論した。

　第十章において本研究の総括を行うととも

に、メタンの芳香族化反応の反応機構とMo／

HZSM・5及びRe1HZSM・5触媒は示す複合触
媒作用についてその特徴を明らかにした。
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Studies　on　HZSM－5－Supported　Mo　and　Re　CataRysts

＿Catalytic　Performances　in　Methane　Aroma伽tion　Reaction

　　　　　　　　　　　　　and　Their　Structural　Characterization一一一

Abstract

　The　main　achievements　in　the　present　studies　are　summarized　as　4　key　points：　1）　HZSM－5

supported　Mo　and　Re　catalysts　are　found　to　be　quite　active　and　selective　for　directly　conyerting

methane　to　benzene，　naphthalene　and　C2　hydrocarbons．　A　great　progress　for　methane

aromatization　has　been　m　ade　because　of　the　discover　of　the　two　new　catalysts．　2）　Coke

deposition　on　the　catalyst　for　non－oxidative　conversion　of　methane　is　solved　by　varying

methane　pressure　combined　with　addition　of　CO2　in　methane　feed．　Coke　selectivity　is　about　O

at　3　atm　methane　and　with　1－30／o　CO2　addition　in　methane　feed．　3）　T　vvo　reactions　on

Re／HZSM－5　catalyst　are　identifried：　i）　methane　with　CO　at　the　nearly　same　consumption　rate

to　produce　benzene，　naphthalene　and　C2　hydrocarbonss　ii）　an　improved　reforming　reaction　o　f

methane　with　CO2　to　produce　aromatics　and　synthesis　gas．　4）　Mo　O3／HZSM－5　and

Re207／HZSM－5　are　first　endothermically　converted　to　its　reduced　state　before　production　of

benzene．　Mo　exist　s　in　it　s　carbide　and　is　highly　dispersed　on　the　HZ　SM－5　support．　T　he　result　s

are　presented　in　details　in　9　chapters．

　　It　vvas　found　in　chapter　2　that　Mo　modified　ZSM－5　catalysts　exhibited　excellent　catalytic

activity　and　selectivity　to　benzene　for　the　conversion　of　methane　at　973　K　under　non－

oxidizing　conditions．　H2　and　ethene　may　be　the　primary　pro（fucts　in　the　conversion　of

methane　to　benzene．

　　It　was　found　in　chapter　3　that　the　cat　alytic　act　ivity　and　st　ability　of　Mo／HZSM－5　cari　be

enhanced　by　the　addition　of　Zr　or　W　and　it　is　also　very　sensitive　to　pretreatment　conditions．

Pretreatment　of　catalysts　under　air　atmosphere　at　a　high　temperature　lead　to　an　increase　in

pore　size　of　the　catalysts　and　a　decrease　in　methane　conversion　and　activity　maintenance．

However，　the　catalysts　can　be　stabilized　under　the　atmosphere　of　methane．　A　new　Mo　phase　is

formed　under　the　methane　atmosphere，　which　is　a　perquisite　of　the　formation　of　arom　atics　and

C2　hydrocarbons．

　　In　chapter　4，　the　TGIDTAIM　S　，　XANESIEXAFS　and　SEMITEM　studies　reveal　that　HZSM－5

supported　Mo　oxide　is　endothermally　converted　with　methane　around　955　K　to　molybdenum

carbide　（Mo2C）　cluster　（Mo－C，　CN．＝＝1，　R＝2．09　A；　Mo－Mo，　C．N．＝2．3－3．5；　R＝　2．98　A）．　The

婦阿一｝

．灘灘…灘懸購響鱗灘騨灘難麟懸羅鑛醗灘餐．　



formed　Mo　carbide　exist　s　on　HZSM－5　support　in　the　uniform　particles．　T　he　highly　dispersed

meta1　carbide　on　HZSM－5　support　acts　as　the　active　center　to　activate　methane　prim　arily　t　o

C2　hydrocarbons，　which　are　secondarily　dehydrocondensed　tovvards　benzene　and　naphthalene

on　HZSM－5　support．

　　Methane　aromatization　with　addition　of　C2　hydrocarbons　such　as　ethane　and　ethene　and

acetylene　on　Mo／HZSM－5　catalyst　is　studied　in　chapter　5．　Addition　of　all　C2　hydrocarbons　in

methane　feed　lead　to　an　increase　of　benzene　form　ation　rate　and　ethane　addition　is　most　effec　t

to　enhance　the　rate　of　benzene　formation．　A　much　high　form　ation　rate　and　selectivity　of

benzene　is　achieved　by　addition　of　about　4．30／o　ethane　in　methane　feed．　The　formation　rate

and　selectivity　of　naphthalene　is　most　effect　ively　enhanced　by　addit　ion　of　4．30／o　acetylene．

The　increase　of　benzene　form　ation　rate　is　due　to　high　concentration　of　［C2Hy］　intermediates

for　benzene　fo　rm　ation　by　addition　of　C2　．　Addition　of　acetylene　lead　to　enh　ancement　of　the

secondary　reaction　of　benzene　to　produce　naphthalene，　so　the　formation　rate　and　selectivity

of　naphthalene　is　high．

　　In　chapter　6，　catalytic　dehydroaromatization　of　methane　with　CO　and／or　CO2　toward

benzene　and　naphthalene　on　Mo／HZSM－5　and　Fe／Co　modified　MolHZSM－5　in　the　prolonged

time　of　100　h　is　studied．　The　catalyst　stability　is　enhanced　and　coke　form　ation　is　suppressed　by

adding　CO　and／or　CO2　t　o　the　m　ethane　feed　and　by　varying　methane　pressure．　Bifunctional

catalysis　of　Mo／HZSM－5　is　discussed　by　the　mechanism　that　methane　is　activated　on　Mo　sites

to　fonn　CHx　and　C2－species，　and　then　secondarily　to　form　benzene　and　naphthalene．

　　Dependence　of　activity　and　selectivity　for　methane　aromatization　on　Mo／HZSM－5　on　the

reaction　conditions　such　as　pressure　and　specifric　velocity　of　methane，　CO2　addition　at　high　SV

and　reaction　temperature　is　investigated　in　chapter　7．　A　high　and　stable　benzene　formation

rate　of　about　1700　nmole／s／g－cat　is　achieved　for　methane　aromatization　on　Mo／HZSM－5　by

increasing　meth　ane　pressure　frem　1　to　3　atm，　m　ethane　specifric　velocity　from　1440　to　10000

mlth／g－cat　and　react　ion　temperat　ure　from　973　to　1073　K，　and　by　adding　about　2．10／o　CO2　to

methane　feed．　The　rate　of　benzene　formatioh　with　time　on　stream　is　remarkably　stabilized　by

varying　methane　pressure　and　addition　of　2．　I　O／o　CO2　in　m　ethane　feed　and　greatly　enhanced　by

increasing　methane　specifric　velocity　from　1440　to　10000　ml／h／g－cat　and　reaction　temperature

from　973　to　1073　K．　The　selectivity　of　benzene　and　naphthalene　and　coke　is　also　controlled

by　varying　reaction　conditions．　High　methane　specifTic　velocity　lead　to　high　benzene

selectivity　and　low　naphthalene　selectivity，



　　Catalytic　features　of　Re／HZSM－5－based　catalyst　for　aromatization　reaction　of　methane　are

studied　and　compared　with　that　of　Mo／HZ　SM－5　in　chapter　8．　lt　is　found　in　T　PR　of

Re207／HZSM－5　in　methane　flowthat　methane　activation　process　is　initiated　by　reduction　of

Re2071HZSM－5，　which　is　endothermic　and　company　with　production　ofhydrocarbon　oxides

and　loss　of　cat　alyst　vveight．　However，　the　reduct　ion　of　cat　alyst　in　hydrogen　stream　is　a

。x。th・・mi・p・・cess・and　tak・・place　at・mu・h　l・me・temp・・at・・e・W・p・・P・・e　th・t・h・ni皿

carbide　is　formed　by　reductien　of　Re207／HZSM－5　in　m　eth　ane　flow．　The　carbide　is　stable　in　th　e

atmosphere　of　methane　and　catalyze　the　reaction　of　methane　aromatization．　The

temperature　for　formation　of　rhenium　carbide　is　about　100　K　lowerthan　that　for　formation　of

molybclenum　carbide　and　Re／HZSM－5　shows　much　better　low－temperature　activity　than

Mo／HZSM－5－based　catalyst．　To　compare　with　Mo／HZSM－5－based　catalyst，　Re／HZSM－5－based

catalyst　exhibits　a　better　lovver－temperature　activity　and　a　higher　selectivity　to　C2

hydrocarbons．

Catalytic　conversion　of　methane　with　CO　to　aromatics　on　Re／HZSM－5－based　catalyst　are

studied　in　chapter　9．　The　CO　involves　in　the　catalytic　conversion　vvith　methane　to　produce

arom　atics　and　the　catalytic　activity　for　benzene　form　ation，　therefore，　is　greatly　enhanced　by

、曲gCO　t・m・th・an・fee己F血h・㎜・・e，　the　c・nversi・n・f　CO輌th　m・th・ne　al・・1ead　t・

formation　of　some　intermediate　products　such　as　CO2　and　H20，　which　eliminate　coke　on　the

catalyst．　Therefore，　the　coke　deposition　is　decreased　and　catalyst　stability　is　enhanced　by　CO

addition．

　　Effect　of　CO2　addit　ion　on　aromat　ization　of　m　ethane　on　Re／HZ　SM－5－based　cat　aly　st　are

s加died　in　chapter　10．　High　and　stable　rate　of　benzene　fomation　is　achieved　at　a　high　SVof

methane　on　Re／HZSM－5　type　catalyst　by　addition　of　about　1－30／o　CO2　to　methane　feed．　A

little　amount　of　CO2　in　methane　feed　is　very　effective　to　suppress　deposition　of　coke　on　the

catalyst　and　to　enhance　selectivity　ofhydrocarbon　products．　Seleetivity　for　coke　deposition　is

about　O　at　the　presence　of　1－30／o　CO2　in　methane　feed　T　he　react　ion　of　CO2　with　coke　to

produce　CO　and　the　secondary　reaction　of　produced　CO　with　hydrogen　to　produce　CHx　species

are　responsible　for　the　low　coke　selectivity　and　high　hydrocarbon　selectivity．
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Chapter　1．　Generall　introduction

1．1．　Aromatization　oflight　hydrocarbons

　　　Transformation　of　light　alkans　such　as　p　rop　ane　and　butane　into　arornatics　has　been

e対ensively　studied　over　the　past　y　ears　because　of　its　significance　in　both　industrial　and

acedemic　points．　The　aromatic　hydrocarbons　can　be　utilized　as　abooster　for　high　octane

number　gasoline　and　are　fundamental　raw　chemicals　in　petroleum　chemistry．　The　best

cataly　sts　for　light　alkane　aromatization　is　Zn　or／and　Ga　modified　Z　SM－5　zeolites　with

bifunction．　The　acepted　mechanism　for　aromatization　oflight　alkanes　is　as　below，

Conversion　routes　of　light　paraffms　on　ZSM－5　zeollites

Para　rrms

（1）

C2－C　4　Olefins

＞Small　olefins

（S　Cracking

（2）レLarger・・1・fin・（3）〉○黛〔：）

Smaller　paraffTins　N

（6）　H－transfer　reaction

｛1）　Peh－y．dro．genation　（2）　Oligomerization，

（3）　Cyclization　（4）　dehVydrogenatiori

（1）　light　alkane　activation　is　initiated　by　dehydrogenation　to　produce　olefin，

（2）　oligomerization　ofthe　olefin　to　produce　larger　olefms

（3）　cyclization　ofthe　1arger　olefins　to　produce　cyclohexanes　or　cyclohexene

1
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（4）　dehydrogenation　of　cyclohexanes　or　cyclohexenes　to　p　roduce　benzene

　　　　　　A　commerciallized　p　rocess　called　as　cy　colar　p　rocess　has　been　develop　ed　for

aromatization　oflight　alkanes．　M　ethane　is　much　more　abundant　and　cheaper　raw　materials

thane　p　rop　ane　and　butane．　However，　the　cataly　sts　such　as　Zn　or　Ga　modified　Z　SM－5

shows　only　very　low　activity　for　aromatization　of　methane．　Therefore，　to　fmd　new

cataly　sts　for　methane　aromatization　is　highly　disirable　and　much　more　siugi）ifcant　from

both　industrial　and　acedemic　points．

1．2．　lndirect　conversion　of　methane

　　　　Methane　can　be　converted　to　usfule　chemicals　indirectly　and　directly．　For　indirect

conversion：the　fust　step　is　to　p　rep　are　synthesis　gns，　and　the　second　step　is　to　p　roduce

usefu1　chemi　cals　such　as　alcohols，　olefins　and　gasoline　etc　from　thynth　esis　gas　as　below，

　CH4　一）　CO　＋　H2　（　to　prepare　thynthesis　gas）

CO　＋　H2　一一〉　usefu1　chemicals　（　alcohols，　olefins　and　gasoline　etc）

　　　Because　two　step　process　is　required　and　the　indirect◎onversion　need　e対ra◎ost　fbr

P・e吻ng・thynth・・1・gお，　m・than・di・ect・・nversi・n・attract・m・・e　a廿・nti・n，・ecently．

1．3．　Direct　conversion　ofmethan

　　　Direct　conversion　of　methane　is　classified　as　oxidat　ive　conversion・and　non－oxidative

conversion．　Oxidative　conversion　includes　partial　oxidation　and　oxidative　coupling：

　　CH，　＋　O，　一一）　CH，OH

　　2CH，　＋　02　一一〉　C，H，＋　H，O

Non－oxidative　conversion　includes：

Homologation　（two　steps）

　　step　1：　CH4　一〉　CH．＋H2　step　2：　CH．＋H2　一〉　C2＋　hydrocarbons

Ther丁nal　pyrolys五s　and　catalyhc　pyrolysis

　　CH，　一　C6H6＋　C2H2　＋　H2　＋C

　　AIthough　cost　of　preparing　thynthesis　gas　is　saved　for　methane　direct　conyersion，　the

problem　is　that　methane　is　yery　stable　molecular．　For　oxidative　conversion，　the　further

2
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oxidation　of　products　is　unavaidable．　Therefore，　the　selectivity　is　low．　For　non－oxidative

conversion，　usually　high　reaction　temperature　is　required．　High　reaction　temperature　for

methane　nonoxidative　conversion　cause　serious　coke　deposition　problem．　Some　data　for

methane　aromatization　by　different　method　s　are　l　ist　i　n　table　1　．

Tabie　1．　Comparation　of　the　result．s　for　methane　dehydroaromatizatien　by

different　methods

Catalyst　reaction　temperature　conversion

　　　　　　　　　　　　　　　　　　　　　　　　　（o／o）

（K）

selectivity

　　　　（o／o）

aromatics　C2

atomatics　yield　reference

（o／o）

pyrolysis

silica　gel

NaCl／MnO2

1sO／o　Mo／HZSM－5

1oO／，　Re／HZSM－5

1480－1580

1273

1223

1073

1073

13．2

12．0

IO．5

23．4　24．2

912

88．0

4．6

6．5

6－10

4．2－7．2

3．1

1　O．9

9．2

（1）

（2）

（3）

率

＊

References：

（1）　H．L．Mitchell　M，　R．H．　Waghome，　U．　S．　Patent，　4　172　810，　1979

（2）　J．　Oro　and　J．　Han，　Science，153　（1966）　1393

（3）　J．B．　Clari　dge，　M　L．H．　GTeen　et　al，　App．　Catal．　A，　89　（1992）　103

★The　results　are　achieved　in　the　present　work；SV＝10000　m1／h／g－cat，　P＝3蜘，

　　with　addiotion　of　2．10／o　CO7　in　methane　feed
　　　　　　　　　　　　　　　　　　　　　　幽

For　thermal　pyrolysis　of　methane，　1480－1580　K　is　required　to　obtaine　about　6－100／o

benzene．　For　catalytic　pyrolysis　of　methane　on　gel，　the　temperature　is　1273　K　to　obtaine

4．2－7．20／o　aromatics．　M　ethane　mixt　ure　with　oxigen　p　ass　through　NaCl／M　nO2　at　high

3



temp　erature　of　1223　K，　about　3．10／o　aromatics　can　be　obtained．　The　above　methods，　both

selectivity　and　yield　of　aromatics　is　low，　and　reaction　temperature　is　quite　high．

　　　　However　recently，　HZSM－5　supported　Mo　and　Re　two　type　catalysts　are　developed

by　us・Th・datawehav・achi・ved・紬istw・cataly紬・m・than，d，hyd，。a，。ma伽h。n

are　compared　with　the　data　in　reference．　9．2－10．9e／o　aromatics　yiel　is　obtained　at　a

temperature　of1073　K　and　aromatics　selectivity　is　quite　high　（about　900／o）．　We　wi11　discuss

those　two　type　catalysts　and　the　catalytic　reaction　of　methane　on　the　catalysts　in　the

following　9　chapters．
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Chapter　2

HZSM－5　Zk）olite一一Supported　Mo　Catalyst　for　Methane　Aromatization

Reaction

Abstract

　It　has　been　found　that　Mo　modified　Z　SM－5　cataly　sts　exhibited　excellent　cataly　tic

activity　and　select　ivity　for　methane　aromatizat　ion　react　ion　at　973　K　without　oxy　gen．

Benzene　is　the　main　aromatic　p　roduct　for　methane　conversion　on　MoHZ　SM－5　cataly　st．

H2　and　ethene　may　be　the　p　rimary　p　roduct　s　in　the　conversion　of　methane　to　benzene．

Medium　p　ore　structure　of　HZ　SM－5　p　lay　a　imp　ortant　role　for　selective　conversion　of

methane　to　benzene．　in　addition，　HZ　SM－5　sup　p　orted　Mo　cataly　st　is　the　most　active

and　selective　for　methane　aromatizat　ion　reaction　among　the　more　thane　10　tested

transiti　on　metals．

1　lntroduction

　in　the　chemical　industries　concerning　the　utilization　of　the　different　carbon　resources，

aleft　diffricult　p　roblem　is　the　chemical　ut　iliz　ation　of　nat　ural　gas　by　methane　direct

conversion，　which　has　challengod　many　minds　becaus　e　methane　is　a　too　stable　molecule．

Over　the　p　ast　y　ears，　methane　conversion　with　oxy　gon　to　ethy　lene　or　methanol　has　been

extensively　studied．　However，　for　the　oxidative　methane　coup　ling　p　rocess，　the　C2－

hy　drocarbon　selectivity　is　substantially　sup　p　ressed　by　increasing　the　methane

conversion　on　Sm203　and　LilSm203，　as　shown　in　table　1．　The　sit　uation　for　methane

partial　oxidation　to　methanol　on　Mo／SiO2　has　no　chango．　Product　selectivity　at　high

methane　conversion　is　low．　Therefore，　the　industry　interests　in　these　p　rocesses　have

been　limited　so　far．

　　Recently，　increasing　attention　has　been　shift　ed　to　the　methane　convers　ion　without

oxy　gen．　The　free　energy　for　methane　convers　ion　without　oxy　gon　is　p　ositive．　To　avoid

5
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the　limit　ation　in　thermody　namios，　two－step　homologation　react　ion　of　methane　was

designed　as　shown　in　fig　1．　M　ethane　is　first　dehy　drogonated　to　CHx　sp　ecies　（xf－rO－3）

on　the　catalyst　at　a　hiゆtempera加re，　which　are　hydrog¢nated　to　hi帥er　hydrocarbon

such　as　ethane　at　a　lower　tempera加re．　This　is　not　feasible　to　industry　because　of　its

low　p　roduct　ivity．　Recent　ly，　we　are　int　erested　in　the　direct　convers　ion　of　methane　to

benz㎝e，　which　beOome　quite　feasible　by　increasing　temp　era加re．　However，　the　carbon

production　is　thermody　namically　much　more　favorable　than　hy　drocarbon　p　roduction　at

high　temp　erature．　The　p　urp　ose　of　the　p　res　ent　work　is　to　convert　methane　to　benzene

selectively　with　mhibition　of　carbon　dep　osit　ion．　This　is　a　big　challcmgo　not　only　to　the

catalyst　designer　but　also　to　a11　the　chemists．

2　Experiments

　MoiCHZ　SM・一5　was　p　rep　ared　by　imp　regnating　NH4Z　SM－5　（SiO2／AI203＝40，　Toso

Co．）　with　NHh）6M　o70244H20　（Kanto　Chem　Co．）　aqueous　solution，　followed　by　drying

at　383　K　for　4　h　and　calcination　at　773　K　for　4　h．　The　fmal　ratio　of　MoMZ　SM－5　in

weight　was　set　to　be　about　30／o．　HZ　SM－5　sup　p　ort　ed　other　transition　metal　cataly　sts

were　p　rep　ared　by　the　same　method．　The　samp　le　was　fmally　p　ressed，　crushed　and　sorted

into　20－40　mesh．

　　Cataly　tic　tests　were　carried　o　ut　with　about　500　mg　cat　aly　st　p　laced　in　a　fixed　bed

cont　inuous－flow　straight　ty　p　e　quartz　reactor．　Aft　er　cat　aly　st　was　p　ret　reat　ed　by　air

stream　at　873　K　for　30　min　and　then　by　He　stream　at　873　K　for　30　min，　the　stream　was

switched　to　feed　gas　and　cataly　st　bed　was　brought　to　a　selected　temp　erat　ure　for　30　min

and　it　fmally　was　brought　to　the　reaction　temp　erature．　The　cataly　tic　test　was　p　erformed

at　a　methane　p　artial　p　ressure　of　100－500　kPa（1　一5　atm）　and　a　methane　sp　ace　velocity　of

6
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1440　ml／gh　at　NTP．　The　raw　mat　erials　of　methane　is　over　99．990／o　p　ure．　All　the　flow

rate　of　gas　was　controlled　with　mas　s　flow　controllers　（ASAHZI　Keiki　Co．　LTD）．　The

products　were　withdraw　p　eriodically　from　the　outlet　of　reactor　and　analy　z　ed　by　two　on－

1ine　gas　chromatograp　hy　of　Shimadaz　u　oc－9A　and　GC－8A　equip　p　ed　with　a　fiame

ionizat　ion　det　ector　（FID）　and　a　thermal　conduct　ivity　det　ector　（TCD）　resp　ectively．

Hy　drocarbon　p　roducts　were　s　ep　arat　ed　by　a　4mm　×lm　Porap　ak　P　column　and　Ar　，　CH4，

CO，　CO2，　H2　were　analy　z　ed　by　a　4　rnmx　l　m　act　ive　carbon　column．　2．20／o　Ar　in　methan　e

was　emp　loy　ed　as　internal　standard．　Formation　rat　e　of　p　roducts，　methane　conversion　and

product　selectivity　were　calculated　by　an　internal　standard　analy　zing　method］．　Below，

the　method　is　p　resented　briefly．　Even　if　the　total　flow　rate　changed　at　outlet　from　inlet

of　reactor　（FT．t，iintiet，　FT．t．iOU’iet　）．　Flow　rate　of　intemal　standard　gas　（F，t）　doses　not　change．

　F，t　＝FTot．10Utlet　X　X，tOUtlet　＝FT．talinlet　x　xstinlet

Here，　X，t　refer　to　concentration　of　the　intemal　standard　gas．

Formation　rate　of　i　product　（FD　can　be　calculated　as，

Fi＝FT．t．iOutiet　X　XiOutiet　＝Fst　×　xioutiet／x，toutiet

By　the　sarne　way，　methane　conversion　（C．．th．．，　）　can　be　cal　culated　as，

Cmethane　＝＝　（Fmethaneinlet　一FmethaneOutlet）／　Fmethaneinlet　＝　1　一　FmethaneOUtlet／　Frnethaneiiilet

”1－FAr　×　XmethaneOutlet／XArO”tlet　×　F．．than．inlet

　　TPSR　（T　emp　erat　ure　P　rogrammed　Surface　React　ion）　exp　eriment　s　was　p　erformed　with

about　200mg　cat　aly　st　p　laced　in　a　U－shap　e　quartz　reactor．　Fig　1　shows　a　scheme　of

setup　used　in　this　work　for　TPAR　exp　eriments．　The　cat　aly　st　zone　was　heated　in　a　He

stream　to　873　K　and　held　at　873　K　for　30　min　then　cooled　to　room　temperature．

7
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At　room　tempera加re　the　stream　was　switched　to　methane　feed．　Tempera加re－

programnied　methane　react　ion　was　start　ed　from　room　temp　erature　to　973　K　at　l　O　K！tm

and　then　held　at　973　K　for　30　min．　The　p　roducts　such　as　H2　（m／e＝＝2），　CO（m／eF28），

CO2（rn／e＝44），　C2H4（m／e＝26），　C2H6　（m／e＝30），　C6H6　（mite＝78）　was　monitored　with　T　E－

150　multip　le　channel　mas　s　spect　romet　er　and　the　sp　ectra　was　rexx）rded　at　the　s　ame　time．

Reaction　pressure　is　34　kPa．

3　Results　and　discussion

　It　is　well　known　that　the　Zn　or　Ga　modified　HZ　SM－5　have　been　emp　loy　ed　to　the

aromatizat　ion　of　C2－C4　hy　dro　carbons．　However，　they　are　limited　if　ap　p　ly　ing　to　methane

feed．　We　t　ry　to　s　eek　the　cat　aly　st　which　is　act　ive　for　methane　aromat　iz　at　ion　react　ion，

therefore　we　combine　the　transition　metals　with　HZ　SM－5　zeolite　and　test　its　cataly　tic

performance．　We　have　found　that　M　oLHZ　SM－5　is　quite　active　and　selective　for　methane

aromatization　But　the　other　t　ested　metals　s　uch　as　Nb，　W，　Fe，　Co，　Ni，　Cu，　V，　Z　r，　Ti，　Pt，

Ru，　etc．　have　no　or　only　a　littl　e　acti　vity　for　methane　aromati　zati　on　．

　　The　cataly　tic　p　erformance　of　HZ　SM－5　sup　p　orted　M　o　cat　aly　st　and　HZ　SM－5

sup　p　orted　other　met　al　s　uch　as　Nb　is　comp　ared　in　t　able　3．　For　HZ　SM・一5　s　up　p　ort　ed　M　o

cat　aly　st，　hy　drocarbons　are　main　p　roducts　and　rate　of　benzene　format　ion　is　quite　high．

However，　for　Nb／HZ　SM－5　and　HZ　SrV［一5　s　up　p　ort　only　exhibit　a　little　activity　for　this

reaction　and　the　main　p　roduct　is　carbon．　The　contribution　of　Nb　to　the　cataly　tic　activity

can　be　neglected．

The　effect　of　the　s　up　p　ort　ing　mat　eri　als　on　the　cat　aly　tic　p　erfo　rm　an　ce　is　also　st　udi　ed　and

the　results　are　shown　in　this　tahle．　30／oM　o　was　sup　p　orted　on　the　different　sup　p　orting

mat　erials　such　as　HZ　SIVI－5，　A1203，　M　ordenite，　FSM－16，　SiO2，　USY．　We　found　that　only

Mo／HZ　SM－5　is　active　and　selective　for　methane　aromatization．　This　is　due　to　that

8
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HZ　SM－5　zeolite　has　a　straight　charmel　and　medium　p　ore　structure．　The　p　ore　size　is

about　5．5　A，　which　is　very　similar　to　the　molecule　size　of　aromatics．　Therefore，　it　has　a

good　shap　e－selectivity　for　aromatics　formation．　Other　sup　p　orting　materials　such　as

A1203，　Mordenite，　FSM－16，　SiO2，　USY　have　a　macrop　ore　structure　and　their　pore　size

is　much　1argor　than　that　of　Z　SM－5　zeolit　e．　The　largor　p　ore　siz　e　easily　lead　to　serious

coke　disposition．

　　A　Study　by　T　emp　erature　Programmed　Aromat　ization　Reaction　（TPAR）　over

Mo／HZ　SM－5，　Z　n／HZ　SM－5　and　HZ　SM－5　gnve　further　informat　ion　on　the　methane

aromatizat　ion　react　ion　as　shown　in　figs．　3－5．　T　wo　imp　ort　ant　observations　can　be　made

from　the　res　ult　s：　1）　the　start　ing　t　emp　erat　ure　for　benzene　format　ion　over　M　o／HZ　SM－5　is

about　973　K，　which　is　ahout　50　K　lower　than　that　over　Z　n／HZ　SM－5．　HZ　SM－5　z　eolite

did　not　exhibit　any　activity　for　the　aromatizat　ion　under　the　given　exp　eriment　condit　ions．

2）　Ethene　and　H2　can　be　direct　ly　det　ect　ed　from　TPAR　and　both　evolved　over

Mo／HZ　SM－5　at　about　973　K，　Which　is　the　same　temperature　at　which　benzene　evolved．

4　Conclusion

1．　Cataly　t　ic　dehy　drocondensat　ion　of　methane　towards　aromatics　such　as　benzene　and

naphthalene　with　bulky　hy　drogen　p　roduction　was　found　to　p　roceed　by　the　HZ　SM－5

zeolite　supported　Mo　catalyst

2．　Benzene　is　the　main　aromatic　p　roduct　for　methane　conversion　on　Mo／HZ　SM－5

cat　aly　st．　H2　and　ethene　may　be　the　p　rimary　p　roduct　s　in　the　conversion　of　methane　to

benzene．

3．　Medium　pore　structure　of　HZ　SM－5　p　lay　a　imp　ortant　role　for　selective　conversion　of

methane　to　benzene．　HZ　SM－5　sup　p　orted　M　o　cataly　st　is　the　most　active　and　selective

for　methane　aromatization　reaction　among　the　more　thane　10　tested　transiti　on　metals．
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Table　1

Thermodynamics　of　CH4　conversion　with　02

Reaction AGO　at　298K
　　（kJ／mol）

AGo　at　1000K
　（kJ／mo1）

CH｝，　＋　1／202　一一一〉　1／2C2H4＋　H20

CH，　＋02　一〉　CH30H

CH，　＋　202　o　CO2＋　2H20

一　143．7

一　115．4

一　800．8

一　152．7

　－75．8

一　800．8

Table　2

The　results　obtained　for　CH4　conversion　with　02

Reaction Catalyst　Conv．（O／o）　S　el．（O／o）

on，　＋1／20，　01／2　C，H2，　＋H，　O

CHL，　＋02　一一一〉　CII30H

　Sm，03　5．8

Lil　Sm203　41．2

Mo／SiO2　1．0

　　　　　　　8．2

The　data　were　taken　from　references　1　and　2．

1．　K．　Aoki　et．　al，　Catal．　Today，　45　（1998）　29．

2．　H．　D．　Gesser　et　al．，　Catal．　Today，　42　（1998）　183．

　　　　　　　　　　　　io
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Table　3

Effect　of　transition　metals　on　the　catalytic　performance　for

methane　aromatization

Catalyst Conv．　Selectivity（O／o）

（O／e）

　　　　　HC　Coke　CO

Formation　Rate

（nmole／s／g－cat）

C6H6　CioHs　C2

30／oMo／HZSM－5

100／，Nb／HZSM－5

HZSM－5

り
」
〆
0
（
∠

弓
」
7
’
8

97　14

2　　0

0．5　O

47

　5

　7

The　data　were　taken　after　ab　out　60　min　o　n　stream　at　973K　and　3　atm

and　S　V＝　1440　ml／h／g－cat

CH4　一一一一一一tp　C6H6　＋　C　i　oHs　＋　C2　＋　H2

Mo／HZ　SM－5　is　quite

aromatization　reactlon

active　and　selective　for　methane
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　　　　　　　Table　4

　　　　　　　Effect　of　supporting　materials　on　catalytic　performance　for　methane

　　　　　　　aromatization　，

Catalyst Conv．（O／o） 　Selectivity（O／o）

HC　Coke　CO

Formation　Rate

（nmole／s／g£at）

C6H6　CioHs　C2

Mo／HZSM－5

Mo／USY

Mo／SiO2

Mo／FSM一　1　6

Mo／Mordenite

Mo／Al，03

9．4

6．4

5．3

6．8

7．3

7．4

63

P3

36

84

87

88

83

89

1 119

　21

　　4

　14

　　9

　　9

29　25

0　13

0．3　10

0　24

0　23

0　14

The　data　were　taken　after　40　min　on　stream　at　973K　and　l　atm；

Mo　loading　is　3　wt．O／o
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Chapter　3
Activity　and　stability　enhancement　of　MolHZSM－5一一hased　catalysts　for

methane　non－oxidative　transformation　to　aromatios　and　C2
hydrocarbons：　Effect　of　additives　and　pretreatment　conditions

Abstract

　The　effect　of　addit　ive　and　p　retreat　ment　condit　ion　on　the　act　ivity　and　stability　of

Mo／HZ　SM－5－based　cataly　st　for　methane　non－oxidative　transformation　to　aromatics　and

C2　hy　drocarbons　is　investigated．　Activity　and　selectivity　of　Mo／HZ　SM－5－based

cataly　sts　can　be　modified　by　the　addit　ion　of　a　s　econd　met　al　comp　onent，　and　the　cat　aly　t　ic

performance　of　the　cat　aly　st　is　very　sensit　ive　to　the　atmosphere　of　p　retreat　ment　at　high

temp　era加re．　High　methane　conversion　of　over　l　O％and　quite　stable　activity　have　been

obt　ained　at　a　react　ion　t　emp　erat　ure　of　973　K　over　M　o／HZ　SM－5，　M　o－W／HZ　SM－5　an　d

Mo－Zr／HZ　SM－5　cataly　sts　pretreated　by　air　at　a　temp　erature　of　873　K．　Cataly　st

pretreatment　under　air　atmosphere　at　a　high　temp　era加re　may　lead　to　an　mcrease　m　pore

siz　e　of　the　cat　aly　st　s　and　a　decreas　e　in　methane　convers　ion　and　activity　maint　enance．

However，　the　cataly　sts　can　be　stabilized　under　the　atmosphere　of　methane．　A　new　Mo

phas　e　is　forrned　under　the　methane　atmosphere，　which　is　a　p　erquisit　e　of　the　formation　of

aromati　cs　and　C2　hydrocarbons．

Key　words：　M　ethane　non－oxidative　transformation，　Mo／HZ　SM－5－based　cataly　sts，

additives，　pretreatment　conditions，　aromatics　and　C2　hy　drocarbons．

1．　introduction

Direct　transformation　of　methane　to　more　valuble　hy　drocarbons　such　as　aromat　ics　or　C2

hy　drocarbons　is　a　facinating　way　for　reasonable　utilization　of　natural　gas　resources．

There　are　t　wo　research　direct　ion　in　this　area’　i）　methane　oxidat　ive　coup　ling　t　o　C2

hy　drocarbons，　ii）　methane　oligomerization　to　aromati　cs．　The　former　has　been　extensively
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studied　over　the　past　for　about　a　decade．　Howev　er，　the　l　atter　one　is　thermodynamically

more　difficult，　especially　in　the　absence　of　oxygen．　The　minimum　temperature　for　the

conv　ersion　of　methane　pyrolyti　cally　to　benzene　i　s　about　1　173　K，　but　the　temperature　for

obtaining　6－10　wt．％　of　benzene　is　1480－1580　K　P］．　Numerous　attempts　have　been　made

to　catal　yze　these　reactions　at　lower　and　more　feasible　temperature．　l　n　1966，　the　synthesi　s

of　aromatic　hydrocarbons　from　methane　with　4．2－7．2％　aromatics　yield　over　silica　gel　at

1273　K　was　reported　［2］．　ln　addition，　methane　aromatization　reaction　over　other　catalyst

systems　such　as　boron　nitride　（BN）　［3］，　multifunctional　multimetal　oxides　［1．4］，　high

silica　zeol　i　te　supported　Pt－Cr　catal　ysts　［S，　meta1　oxide　and　supported　meta1　catal　ysts　［6］，

high　Ga－contained　Pt／H－Ga－silicates　M，　Mo／　HZSM－5　and　Zn！HZSM－5　catalysts　［8］，

carbonaceous　materi　al　s　［9］　has　been　investi　gated　since　1980．　More　recentl　y，　the　effect　of

W　additive　on　the　catalytic　activity　and　coking　rate　of　Mo／HZSM－5　catalyst　for　methane

and　ethane　aromati　zation　has　been　reported　．［10］．　Methane　acti　vati　on　and　conversion　over

Mo　modi　fied　aci　dic　zeol　i　te　catal　yst　can　proceed　without　the　need　of　any　oxi　dants．

Compared　with　methane　oxidative　coupl　ing　（MOC）　and　methane　pani　al　oxtdati　on

reactions，　the　ad　v　antages　of　the　methane　non－oxidative　transforrnati　on　process　i　s　that　the

sel　ecti　v　i　ty　to　higher　hydrocarbon　products　i　s　hi　gh　and　the　main　product，　benzene，　i　s

easier　to　be　separated　from　the　methane　feed．　l　n　additi　on，　many　problems　causing　from

the　side　reaction　such　as　methane　complete　oxtdation　caii　be　avoided　under　non－oxi　dative

reaction　conditions．　Recentl　y，　the　study　of　methane　non－oxi　dati　ve　transformation　on　Mo

modi　fi　ed　zeolite　catalysts　attracts　more　attention．　An　i　ncreasing　number　of　papers　on　thi　s

research　area　have　been　published　recentl　y　［11－16］，　i　n　whi　ch　characterization　of　catal　yti　c

and　physical　properti　es　of　the　catal　ysts　has　been　reported．　However，　the　reported　catal　ytic

reaction　resuks　of　methane　ov　er　Mo－containing　HZSM－5，　especial　ly　the　results　on　the

acti　v　ity　and　stabi　l　i　ty　of　catal　ysts　are　v　ery　different　since　they　depend　on　the　preparation

and　pretreatment　conditions　of　the　catalysts．　Enhancement　of　the　catal　yti　c　acti　vity　and

stabiliry　of　the　catal　ysts　is　significantly　important　for　obtaini　ng　a　commercialized　process

for　this　reaction．　ln　order　to　obtain　a　highly　active　and　stable　catal　yst　for　the　reaction，　the

17



M1NNI1Ml

，

effect　of　pretreatment　condi　ti　ons　and　additives　on　the　catalytic　performance　of　Mo／［HZSM－

5－based　catal　ysts　are　presented　here．

2．　Experi　ment

2．1．　Catal　yst　preparati　on

Commercially　available　ZSM－5　zeolite　（SiO2／A12C［3＝50）　was　first　converted　into　the

ammonium　form　（NH4ZSM－5）　by　repeated　ion　exchange　（4　times）　with　I　N　NH4NCB

aqueous　solution　at　around　368　K　for　1　h．　l　t　was　then　dried　at　about　383　K　for　4　h．

HZSM－5　and　Mo／HZSM－5　were　prepared　as　descri　bed　previously　［8］　；　Mo－Zr／HZSM－5

was　prepared　by　consecutivel　y　impregnating　NH4ZSM－5　with　zirconi　um　nitrate　and

ammonium　heptamolybdate　aqueous　sol　ution；　Mo－W／HZSM－5　was　prepared　by　co－

impregnating　NH4ZSM－5　with　ammonium　molybdate　and　ammonium　tungstate　aqueous

solution；　Mo－VLHZSM－5　was　prepared　by　co－impregnating　NH4ZSM－5　with　ammonium

molybdate　and　ammonium　yanadate　aqueous　sol　ution；　Mo－LelHZSM－5　was　prepared　by

consecutiyely　impregnating　NH4ZSM－5　with　lanthanum　nitrate　and　ammonium

heptamolybdate　aqueous　solution．　They　were　dried　at　383　K　for　4　h　and　calcined　at　773

K　for　4　h．　The　Mo　loading　of　these　samples　is　2％．　Finally，　the　catalysts　were　pressed，

crushed　and　sorted　i　nto　．20－40’　mesh．

2．2．　Catalytic　tests

Catalytic　tests　were　carried　out　with　about　500　mg　catalyst　placed　i　n　a　fix，　ed　bed

continuous－flow　quartz　reactor．　The　catalysts　were　first　pretreated　under　air　or　，He　stream

in　the　temperature　range　of　773－973　K　for　30　min．，　then　the　stream　was　ssvitched　to

methane　and　the　reactor　was　brought　to　the　reaction　temperature．　A　l　l　the　catal　ytic　tests　，

svere　performed　at　a　pressure　of　200　kPa　and　a　methane　space　veloci　ty　of　1440　ml　g一一1

h一一1．　The　products　svere　withdrawn　periodical　ly　from　the　outlet　of　the　reactor　and
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analyzed　by　an　on－line　Shimadzu　GC－9A　gas　chromatography　with　a　2　m　long　Porapak　Q

column　and　a　hydrogen　frame　ionization　detector．　Conversion　and　sel　ecti　v　ity　were

cal　culated　on　the　carbon　number　basi　s．

2．3．　Coke　deposition　and　BET　surface　area　measurements

Specific　surface　area　and　pore　size　were　deterrnined　by　the　B　ET　method　with

Micrometrics　ASAP－2000　equi　pment　using　nitrogen　as　analysis　gas．　The　amount　of　coke

deposited　on　the　catalysts　was　determined　by　TG－DTA　method．

2．4．　Temperature－Programmed　Surface　Reactions　（TPSR）

The　experi　ment　was　performed　with　about　200　m　g　catal　yst　pl　aced　i　n　a　U－shape　q　uartz

reactor．　Firstly，　the　catalyst　zone　was　pretreated　under　a　He　stream　at　873　K　for　30　min

and　then　brought　to　room　temperature．　A　t　room　tem　perature，　the　stream　was　switched　to

methane．　Temperature－programmed　m　ethane　reacti　on　was　started　from　room　temperature

to　973　K　at　10　K　min一　1　and　then　held　at　973　K　for　30　min．　The　product　such　as　CO

（m／e＝28），　CO2　（m／e＝44），　C2H4　（m／e：一＝26），　C2H6　（m／　e＝30），　C6H6　（m／e＝78）　was

monitored　with　TE－150　multiple　charmel　mass　spectrometer．　The　reaction　pressure　i　s　34

kPa．

2．5．　Materi　al

Methane　feed　i　s　over　99．99qo　pure．

3．　Results　and　discussion

3．　1．　Effect　of　additi　ve　and　pretreat，ment　atmosphere
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Tab］e　1　shows　the　effect　of　additives　on　the　reacti　v　ity　of　methane　over　Mo／　HZSM－5－

based　catalysts．　The　results　in　Table　1　indicate　that　the　acti　v　ity　and　se］ecti　v　ity　to　aromatics

can　be　improved　by　the　addition　of　Zr　or　W　to　MO／HZSM－5．　The　addition　of　Zr　is　most

effective　in　improving　the　activity　of　catalyst　at　a　reaction　temperature　of　9：2t3　K　and　a　Zr

content　of　1　．090　．　The　addition　of　V　Leads　to　a　decrease　in　methane　conversion　and

selecti　v　i　ty　to　aromatics，　but　the　selectivity　to　C2　hydrocarbons　is　sl　i　ghtly　i　ncreased．　The

effect　of　catalyst　pretreatment　atmosphere　on　the　catal　ytic　acti　v　i　ty　of　Mo－Zr／HZSM－5　i　s

summarized　i　n　Tabl　e　2．　lt　i　s　clear　from　Tab］e　2　that　the　catal　ytic　acti　v　ity　of　Mo－Zr／HZSM－

5　can　be　enhanoed　when　i　t　i　s　pretreated　under　fl　owing　ai　r　at　a　temperature　of　873　K．　The

effect　of　pretreatment　conditi　ons　on　specific　surface　area　and　pore　size　of　the　catal　yst　i　s

al　so　examined．　The　results　are　summarized　in　Table　3．　The　treatment　of　Mo一・Zr／HZSM－5

under　an　ai　r　atmos　phere　and　a　high　tem　perature　of　973　K　Leads　to　a　decrease　i　n　B　ET

speci　fi　c　surface　area　and　an　enlargement　of　average　pore　diameter　of　the　catalyst．　The

BET　speci　fic　s　urface　area　and　pore　v　olume　of　the　catal　yst　are　ai　so　decreased　with　ti　m　e　on

stream　at　a　reacti　on　tem　perature　of　973　K，　whi　ch　res　ul　t　from　coke　depositi　on　on　the

catal　yst．　The　B　ET　s　urface　area　and　pore　v　olume　can　be　completely　recov　ered　after　the

coke　is　remov　ed　by　combusti　on　with　ai　r　at　a　temperature　of　773　K．　Therefore，　the　effect

of　air　and　methane　atm　ospheres　on　the　speci　fi　c　surface　area　and　pore　v　olume　of　the

cataJ　yst　is　very　different．

3．2．　Effect　of　pretreatrnent　tem　perature

The　effect　of　pretreatmen　t　tem　perature　on　the　catal　ytic　performance　of　M（》Zr／HZSM－5

cata］yst　i　s　compared　in　Figs．　1－3．　Mo－Zr／HZSM－5　catalysts　pretreated　under　fiowing　air

at　973，　97；B　and　873　K　are　descri　bed　as　Mo－Zr／HZ－973，　Mo－Zr／　HZ－97“3　and　Mo－Zr／HZ－

873，　respectively．　Methane　conversion　and　activity　maintenance　are　very　sensitive　to　the

pretreatment　temperature　of　the　catal　yst．　Mo－Zr／　HZ－873　exhibits　quite　high　and　stabl　e

acti　vity　and　selectivity　to　aromatics　with　time　on　stream．　The　methane　conversion　is　over

10qo　with　selectivity　of　more　than　93　90　to　aromatics　and　about　6％　to　C2　hydrocarbons．
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The　ainount　of　coke　deposited　on　the　catal　ysts　is　less　than　O．25qo　of　methane　feed．　The

cataユyti　c　activity　and　stabi　lity　of　Mo－Zr／HZS　M－5　for　methane　aromati　zati　on　are　evidentl　y

decreased　with　i　ncreasing　pretreatment　temperature　of　the　catal　yst　as　shown　i　n　Fi　gs．　2

and　3．　Therefore，　the　pretreatment　conditi　ons　are　crucial　for　obtai　ning　hi　ghl　y　acti　v　e　and

stable　cata］yst．　In　order　to　compare　Mo／HZSM－5　with　Mo－Zr／HZSM－5　and　MeW／

HZSM－5，　methane　conversion　and　selecti　v　ity　to　aromatics　and　C2　hydrocarbons　as　a

function　of　time　on　stream　over　Mo－WIHZSM－5　and　Mo／HZSM－5　are　shown　in　Fi　gs．　4

and　5．　Enhancement　of　catalytic　activi’　ty　maintenance　by　Zr　or　W　is　also　clear　from　Figs．

4　and　5．

3．　3．　Nature　of　the　active　center

The　pure　methane　reacti　on　process　on　Mo－Zr／HZSM－5　at　elevated　temperature　is　shown

in　Fi　g．　6．　I　t　i　s　clear　from　Fi　g．　6　that　methane　reaction　on　Mo－ZriCHZSM－5　initiates　with

CCP．．　and　CO　formation．　C6H6，　C2H4　and　C2H6　are　then　formed　after　the　disappearance

of　CO．＊　products．　The　formation　of　carbon　oxides　cannot　be　simpl　y　ascri　bed　to　the

reduction　of　Moo3　by　methane．　A　sharp　peak　of　CO　product　i　s　still　appeared　in　the　case

of　Mo／HZSM－5　catalyst　after　reduction　under　H2　stream　at　973　K　as　shown　in　Fi　g．　7．　lt

is　suggested　that　Mo　phase　transformation　takes　place　at　the　first　step　of　methane　reaction

on　the　catalysts，　which　is　a　prerequisite　of　aromatics　and　C2　hydrocarbon　formation．　The

new　Mo　phase　most　possibly　is　molybdenum　carbide　（MoC）．　The　formation　of

molybdenum　carbide　active　phase　for　methane　non－o；　idative　transformation　to　aromatics

and　C2　hydrocarbons，　therefore，　is　suggested　as　folloWS，　cH．　一一．　c　＋2H，

MoO3　十C．　MoC　＋　CO2　＝，　CO　MoO2　十3C　一　MoC　十　2CO　On　the　active

phase，　methane　non－oxi　dative　transforrnation　to　higher　hydrocarbons　oocurred．　The　air

pretreatment　of　the　catalyst　at　a　suitable　temperature　may　induce　the　formation　of

molybdenum　oxide　svhich　fayors　the　formation　of　carbide　active　phase．　The　activity　of　the

catal　yst，　therefore，　is　ev　idently　enhanced　by　air　pretreatment　as　shosvn　in　Table　2．
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Mechanism　for　methane　acti　v　ati　on　and　transformation　to　aromati　cs　and　C2　hydrocarbons

has　been　discussed　in　our　previous　papers　［8，10，12］．

4．　Conclusions

The　catal　ytic　activity　and　stability　of　Mo／HZSM－5　can　be　enhanced　by　the　addition　of　Zr

or　W，　and　it　is　also　v　ery　sensitive　to　pretreatment　conditions．　Hi　gh　methane　conversion　of

over　10％　and　qui　te　stabl　e　acti　v　i　ty　can　be　obtaj　ned　at　a　reacti　on　tem　perature　of　973　K　o　v　er

the　catal　ysts　pretreated　by　air　at　a　temperature　of　873　K．　Pretreati　n　g　catal　ysts　under　ai　r

atmosphere　at　high　temperature　may　l　ead　to　an　i　ncrease　in　pore　size　of　the　catalysts　and　a

decrease　in　methane　conversion　and　activity　maintenance．　However，　the　catalyst　can　be

stabilized　under　the　atmosphere　of　methane．　A　new　Mo　phase　is　formed　under　methane

atmosphere，　which　is　a　prerequisite　of　the　formation　of　aromatics　C2　hydrocarbons．
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Table　1

Effect　of　additives　on　the　reactivity　of　methane　over　Mo／HZSM－5－based　catalysts

Additives　Additive　content（90） Conversion　（90） Selectivity　（90）

C2 Benzene Toluene B＋T　（Ar）

U
W
v
み
み
み

　
0
～
　
0
5
0
0

　
リ
　
　
　
ロ
　
　
　
ロ
　
　
　
ロ
　
　
　
　
　
　
　
ロ

O
1
0
1
∩
V
－
へ
∠

3．6

3．3

4．0

2．7

3．1

5．2

3．4

8．3

6．1

5，0

1L1
6．5

5．8

6．5

86．　1

87．9

8．7．5

77．8

87．1

86．5

79．4

5．6

6．0

7．5

1　L1

6．4

7．7

4．1

91．7

93．9

95．0

88．9

93．5

94．2

93．5

reaction　temperature：　923　K

Table　2

Effec【of　pretreatment　conditions　on　the　catalytic阿fomance　of　Mo・Zr／HZSM・5

Pretreatment　conditions Conversion　（90）　Selectivity

Temperature　（K） Atmosphere C2 Benzene Toluene B＋T　（Ar）

　923

　923

　873

　873

　773

．　773

air

He

air

He

air

He

5．2

2．6

5．5

2．0

42
2．6

5，8

4．0

4．0

10．0

10．0

1　2．7

90．9

88．5

86．5

85．0

70．2

80．8

3．3

7．5

9．5

5．0

19．8

6．5

94．2

96．0

96．0

90．0

90．0

87．3

GHSV＝1440　ml　g－ih－i；reiction　pressure＝200” oa；reaction　temperature＝923　K・

Table　3

Effect　of　pretreatment　atmosphere　at　high　temperature　on　the　BET　surface　area　and　pore　volume　of　Mo－Zr／

HZSM－5　ca：alyst

Treatment　conditions BET　surface　Pore　volume　（cm3　g一：）

area　（m2　g’i）

Micropore Tota1

Average　pore

diameter　（A）

Fresh

Methane　atmosphere

973　K

Air　atmosphere

973　K

　　　　　　　　　　　　　　　130

with　coke　87
after　combustion　138

0f　coke

　　　　　　　　　　　　　　96

O．04227

0．0314

0．0443

0．0313

O．0669

0．0461

0．0687

0，0564

く
ノ
∩
V
《
ノ
　
6
6

響
墨
う
ん
一
　
1
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Chapter　4

Characterization　o　f　Mo／HZS　M－5　c　ataly　st　by　T　G－DT　A－ms，　XAF　S，　EXAF　S

SEM　and　TEM

Abstract

Mo／HZSM－5　catalyst　is　characterized　by　TG－DTA－MS，　XAFS，　EXAFS，　SEM　and

TEM　techniques．　TG－DTA－MS　experiments　indicate　that　the　Mo　oxide　is　reduced　by

methane　to　its　carbide　at　about　962　K　on　HZ　SM－5　sup　p　ort，　which　is　a　strong

endothermic　process　and　accomp　any　with　weight　loss　of　cataly　st　and　production　of　CO，

H20　and　H2．　Only　after　the　melybdenum　carbide　is　formed，　the　hy　drocarbon　products

such　as　benzene，　naphthalene　and　ethene　can　be　formed．　This　strongly　sugots　that　the

Mo　carbide　is　the　active　center　of　the　Mo／HZ　SM－5　cataly　st　for　methane　aromatization．

The　XANESIEXAFS　studies　reveal　that　the　zeolite　supported　Mo　oxide　is　converted

with　methane　to　moly　bdenum　carbide　（Mo2C）　cluster　（Mo－C，　C．N．　＝1，　R＝　2．09；　Mo－

Mo，　C．N．＝2．3－3．5；　R　2．98　A）．　T　he　formed　M　o　carbide　exists　on　HZ　SM－5　sup　p　ort　m

the　uniform　p　articles．　The　highly　dispersed　metal　carbide　on　HZ　SM－5　sup　p　ort　acts　as

the　act　ive　cent　er　to　act　ivat　e　methane　p　rimarily　to　C2　hy　drocarbons，　which　are

secondarily　dehydrocondensed　towards　benzene　and　naphthalene　on　HZ　SM－5　s　upport

1　lntroduction

　Since　Mo／HZSM－5　catalyst　for　methane　aromatization　was　first　reported　in　1993　［1］，

many　promising　results　have　been　achieved　on　this　catalyst．　ln　order　to　understand　the
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mechanism　for　methane　aromatization　on　this　catalyst　and　the　structure　of　the　catalyst，

we　have　conducted　TG－DTA－MS，　XAFS／EXAFS，　SErmM　experiments　in　th　i　s

chapter．

2　Experiments

2．1．　DTA－TG－MS　experiments

DTA－TG－MS　experiments　were　performed　with　a　DTA－TG－MS　combined

instrument　（IV［as　Science　DTA・・2020S　＋　Thermolab）．　Fig．　1　briefiy　shows　the　scheme

o価ese加p　for　TG－DTA－MS　eXperimentS．　About　40　mg　catalyst　was　paced　in　a

alumina　container　under　flowing　l　50　ml／n血He　and　15　ml／㎡n　methane　with　ramping

th　e　temperature　of　10　K／min．　The　weight　change　of　catalyst　was　recorded　by　TG，　h　eat

effect　of　the　reaction　was　recorded　by　DTA　and　the　products　for　the　reaodon　were

monitOred　by　MS　at　the　same　time　du血g　the　catalyst　was　heated　from　room

temperature　to　973K　by　a　programmed　temp　erature．

2．2．　XAFSNXAFS　experiments　and　analysis

　　The　powdered　sarnples　were　charged　under　nitrogen　in　an　in－situ　XAF　S　cell　with　a

Kapt・n丘lm舳d・w（O．5mg）t・P・event　e・rP・s肛e・f血e　sample　t・ai・・　M・K－edge

XAFS（X一・ay・ahs・rPti・n血e　s甘u伽・e）measu・emen偲were　c・nducted・n止e　sample・f

Mo／HZ　SM－5　after　impregnation，　cal　cination　at　873　K，　and　reaction　with　methane　at　873

K　and　973　K　at　BL－10B　at　the　photon　factory　of　the　National　Laboratory　for　high

Energy　Physics（KEK－PF，Tsukuba，・Japan）．　The・ene・gy　and止e・current・f血e　el・伽n

（or　p　osition）　were　2．5　GeV　and　250　mA，　respectively．　A　Si（311）　channel－cut

monochromator　was　us　ed．　The　XANE　S　（X－ray　ab　sorption　near　edge　structure）　and

EXAFS　（extended　X・一ray　absorption　fme　structure）　was　analyzed　by　a　computer

program　supplied　by　Technose　Co．　LTD　［　2　］．　k3　一weighted　EXAFS　function　was

Fourier　transformed　into　R－space　using　the　k－range　from　3。5－18A’1．　The　Hanr血g
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fu）ction　used　was　6＝＝O．5　A’i　．　’lhe　phase　shift　was　not　corrected　for　the　preliminary

Fouri　er　transformation．　The　inverse　Fouri　er　transform　was　cal　cul　ated　to　obtain　a　filtered

EXAF　S　fimction．　The　R　range　of　an　inverse　Fourier　transformation　was　taken　from

1．09　to　3．31　A．　The　Hanning　fimction　of　A＝＝O．05　A　was　used　as　a　window　function．

The丘tting　EXAFS　parameters・f血e・sample・we・e・determined・f・・。…di面・n　number

（C．　N．）　and　interatomic　distance　R．　to　minimize　6　and　correction　of　threshold　energy

（△E。）．The・backscatt・面9　ampli加des㎝d　phおe　shi丘㎞αi・ns・fM・一M・・M・一C・

㎝dM。一〇b。n（㎞騨e　c・rrected　using止・se・fM・f・il（M・一M・）・M・・C（M・一M・・

Mo－C），　and　MoO3　（Mo－Mo），　Mo－O）．　The　EXAFS　parameters　for　discussion　were

五x。d・ar。皿d血e・ptim㎜仙e，　and・the・residual・fact・r・R・was・calculated・t・・ffe・the

。ptimum曲脚継e・th・・p訂amet・・s．乃e・e醐ue　wおes廿m飢ed曲g　a・esidual

factor　R　which　i　s　smaller　than　twi　ce　the　optimum　value．

3　Resuks　and　discussion

It・is・eVid，ntly　indicated　fr・m　TG－DTA－MS　spe伽in丘92血舳e　M・・Xide　is

行耐ly　reduced　by　methane　t・its（励ide舳・ut　962　K・n　HZSM－5　supP・貢・The

phase　transformation　process　from　molybdenum　oxide　to　its　caibide　is　a　strong

endothermic　process　as　shown　by　DTA　profile，　which　accompany　with　weight　loss　of

ca伽ys幡sh・wn　by　TG　p・・丘le，　and・CO，　H・O・and　H・訂e　pr・duced飢血e　same　time　as

monitored　by　Mass　Spectra　Only　after　the　molybdenum　carbide　is　formea　the

hydrocarbon　products　such　as　benzene，　naphthalene　and　ethene　cun　be　formed　This

strongly　suggests　that　the　Mo　carbide　is　the　active　center　ofthe　Mo／HZSM－5　catalyst

for　methane　aromatization．　The　reduction　of　Mo　oxide　by　methane　to　its　caibide　is

quite　different　from　its　reduction　by　hydrogen　to　Mo　metal．　The　TG－DTA－MS　spectra

。fM。／HZSM－5　in　hyd・・gen且・w　in　fig．　3　indicate血at血e・e　is　n・heat　effect　can　be

identified　by　DTA　profile　for　reduction　process　of　Mo　oxide　by　hydrogen，　although　the

catalyst　weight　loss　is　identified　as　by　TG　profile　and　H20　product　is　monitored　by

E’



MS．

　　in　order　t　o　obt　ain　more　direct　evidence　for　the　format　ion　of　moly　b　denum　carbide，

the　XANES厭AFS　eXPeriments　were。・nduCted．　Figure　4　sh・ws　s・me　XANES

sp瓠・a・f6瓢％1・ぬgM・贋ZSM－5副。圃，　t・姻M・wing　methane　at　973　K　fo「

1　and　24h．　The　spectra　of　MoO3　，　Mo　foil　and　a　Mo2C　powder　were　observed　as

references．　All　the　sp　ectra　（figs．　4c　and　4d）　of　Mo／HLZSM－5　after　methane　reaction　at

973Ksh。wed・id，nti戯⑱P・siti・n，　similar・t・M・（；arbide（丘94e）and　M・f・il（fig

4躰the丘・st　infleCti・n　p・int・in・the・apidly　rising　P・貢i・n・fthe　edgO・This　p・sltlon

was　assigned　an　energy　of　20003．9　eV．　lt　is　known　that　edgo　features　move　to　lower

energy　as　the　oxidation　state　ofMo　is　reduced　［3，4　］，　as　shown　for　Mo／CHZ　SM－5　sample

after　the　reaction　with　methane　at　973　K，　compared　with　those　of　the　calcined　sample

（fig　4b）　and　MoO3　（fig　4a）．　The　only　notable　difference　between　Mo　foil　and　Mo2C

XANES　sp瓠ra　wおase噸i・n・f　th・tw・br・ゆeaks　at・20・04　KeV　and　20・03　KeV

above　the　absorption　edgo．　The　XANES　spectrum　of　Mo　foil　showed　a　1argor

separation　than　the　Mo2C　spectrum　The　XAFS　ofthe　Mo／HZSM－5　sample　reacted

with　methane　at　973　K　for　1　and　24　h　（figs．　4c　and　4d）　look　basically　identical　to　the

Mo2C　reference　spectrum　（fig．　6e）　not　to　the　Mo　foil　（fig．　4b．

The　FT　K－edgo　EXAFS　spectra　in　fig　5　indicate　that　the　Mo－Mo　distance　for　Mo

oxide　is　between　3　and　4　A．　Aft　er　the　cal　aly　st　exp　osed　to　methane　at　973　K，　it　changes

t。between　2　、nd　3　A，　whi・h　i・at・th・・ex・Ctly　same　p・siti・nおthe　M・一M・出・伽ce・f

M。2C，eference　sample　and　is　at　the・quite戯・ent　p・siti・n丘・m　th飢・f　M・f・il

sample．　This　results　have　further　improved　that　molybdenum　oxide　is　reduced　by

methane　to　its　carbide　and　not　to　Mo　metal　on　HZ　SM－5　support　after　reaction　with

methane．　HZSM－5　．supported　Mo　carbide　is　the　active　center　for　methane

aromatizatton　reaction．

By　fitting　the　EXAFS　spectra　s・me　st煎u・e・paramete・s　have　been・btained　and

listed　in　table　1．　Table　1　indicates　that　the　Mo－M　o　distance　of　the　formed　Mo　carbide　is

same　as　bulky　moiybdenum　carbide，　although　it　is　supported　on　HZSM・一5　zeolite．
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However，　the　coordnation　number　of　supported　Mo　carbide　is　2．1－3．5　is　much　lower

than　that　of　bulky　Mo　carbide，　which　is　12．

　　The・surface・m。叩h・1・gy・fthe・f・・med・m・lybdenum曜bide　supP・rt　ed・n　HZSM－5

was　observed　by　scanning　electronic　microscope　（SEM）　as　shown　in　fig　6．　After

methane　react　ion，　the　clear　moly　bdenum　carbide　can　b　e　observed　by　SEM　imagine，

which　exists　in　the　uniform　p　art　icle　on　HZ　SM－5　s　up　p　ort．　However，　before　the　react　ion

of　methane，　no　any　particles　but　only　zeolite　crystal　can　be　observed．　This　indicates　that

the　Mo　oxide　is　highly　disp　ersed　on　the　zeolite　after　cal　cinadon．

　　The　cross－section　of　the　cataly　st　is　observed　by　transmission　electronic　microscop　e

（ITEM）　as　shown　in　fig　7．　lt　is　indicated　in　fig　7　that　theparticles　is　mainly　dispersed

on　the　surface　of　the　zeolite　cry　stal．

　　　　Based　on　the　above　results，　we　have　p　rop　osed　a　mechanism　for　methane

aromatization　on　Mo／HZSM－5，　which　is　schemed　as　fig　8．　The　first　step　for　the

reaction　is　that　Mo　oxide　supported　on　HZ　SM－5　was　reduced　by　methane　to　its　carbide，

and　methane　was　oxidized　to　CO，　H20　and　H2．　HZSM－5　supported　Mo　carbide　is　the

active　center　for　methane　activation．　On　the　formed　Mo　carbide，　methane　are　first

dehyd，。genated　t・CHx　sp　ecies，　whi・h・are　dimerized・t・C・且y　species　such　as　ethene・

C2H4　is　secondarily　converted　to　aromatics　such　as　benzene，　naphthalene　even　coke　on

HZSM－5　neolite．

4　Conclusien

The　TGIDTAIMS　，　XANESIEXAFS　and　SEMLrEM　studies　reveal　that　HZSM－5

supP。rted　M・・ぬde　is　end・th・rmally。・nverted　with　methane　ar・皿d　955　K　to

molybdenum　carbide　（Mo2C）　cluster　（Mo－C，　C．N．＝1，　R＝　2．09；　Mo－Mo，　C．N．＝2．3－3．5；

R2．98　A）．．The・f。rm・d　M・carbide・eXists・n　HZSM－5　supP・貢in血e曲m　p面cles・

The　highly　dispersed　metal　carbide　on　HZSM－5　support　acts　as　the　active　center　to

32

　　　　　　　　　　　　
　　コ　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　ヒ　　　　　　　　　　　　　　コ　　　　　コ

1灘1講灘’錨
　．



activat　e　methane　p　rimarily　to　C2　hy　drocarbons，　which　are　s　econdarily

dehydrocondensed　towards　benzene　and　naphthalene　on　HZ　SM－5　support
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Fig．　1．

Schematic　diagram　of　the　setup　for　TG－DTA－MS
experiment
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Fig．　2．

flow
TG－DTA－MS　spectra　o　f　3　wt9／o　Mo／HZS　M－5　in　methane
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Fig　3．

flow

TG－DTA－MS　spectra　o　f　3　wtq／o　MoZHZS　M－5　in　hydro　ge　n
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／rptge’dsge　ExAFs　spectra　of　calcined　and　reacted　3wt％Mo／HZSM－5
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Fig．　6．

SEM　image　of　Mo／HZSM－5　before　and　after　methane　reaction
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Fig．　8．　Mechanism　for　methane　aromatization　on　Mo／HZSM－5
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C㎞裂pter　5

Methane　Aromatization　with　Addition　of　C2　hydrocarbons　on

Mo胚M－5

Abs血俄ct：

　　Methane　aromatization　with　addition　of　C2　hy　drocarbons　such　as　ethane，　ethene　and

acety　lene　on　Mo／HZ　SM－5　cataly　st　is　studied．　Formation　rate　and　selectivity　of

benzene　and　nap　hthalene　is　quite　dep　endent　on　the　addit　ion　of　various　C2　hy　drocarbons

in　methane　feed．　A　ddit　ion　of　a11　C2　hy　drocarbons　lead　to　increas　e　of　b　enz　ene　format　ion

rate　and　ethane　addition　is　most　effect　to　enhance　the　t’ai　e　of　benzene　formation．　A

much　high　formation　rat　e　and　select　ivity　of　benzene　is　achieved　by　addition　of　about

4．30／o　ethane　in　methane　feed．　The　format　ion　rat　e　and　s　electivity　of　nap　hthalene　is

most　effect　ively　enhanced　by　addit　ion　of　4．30／o　acety　l　ene　in　methane　feed．　A　lthough

addit　ion　of　a11　C2　hy　drocarbons　lead　to　increas　e　of　the　act　ive　int　ermediat　es　for　benzene

formation，　the　secondary　reaction　of　benzene　to　p　roduce　nap　hthalene　is　assisted　by

addition　of　only　unsaturated　C2　hydrocarbons，　especially　by　addition　of　acetylene．

Keywords：

methane　aromatization　，　Mo／HZSM－5，　C2　addition，　benzene，　naphthal　ene
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1　lntroduction

　　　　Mo／HLZ　SM－5　cataly　st　was　first　emp　loy　ed　to　non－oxidative　dehy　drogymation　and

aromatization　reaction　of　methane　in　1993　（1］．　The　cataly　tic　tests　were　p　erformed　in　a

continuous－flow　stainless　stee1　react　or　and　in　a　quartz　tubular　microflow　reactor．　The

product　was　analy　zed　by　a　GC　equip　p　ed　with　a　FID　detector　and　by　a　TE－150　mass

sp　ectrometer，　resp　ectively，　but　only　benzene　as　aromatic　p　roduct　was　measured　in　the

exp　eriment　condit　ions．　About　t　wo　y　ears　1回目　er，　more　informat　ion　on　this　react　ion

sy　stem　was　rep　orted　by　other　authors　［2－4］　and　by　us　［5－6］．　Recently，the　unique

cataly　tic　activity　of　moly　bdenum－containing　HZ　SM－5　cataly　st　for　aromatios　p　roduction

（hrectly　from　methane　has　attrac舵d　an　increasing　attention［7－20］．　The　characterization，

prep　arat　ion　an　d　modifi　cat　ion　of　cat　aly　st　an　d　react　ion　kinet　i　cs　have　been　investigat　ed　by

various　techniqueS［7－15］．　The　catalytic　activity　and　stability　of　M　oiCHZ　SIN（（一5　have

been　imp　roved　by　addition　of　different　modifiers　［6，16－19］．　lt　has　also　been　found　that

the　cataly　tic　activity　and　stability　of　cataly　st　is　very　dep　endent　on　p　retreat　ment

conditions　［16］．　M　ore　recently，lchikawa　et　al　［19］and　Lunsford　et　al．　［20］　gave　more

comp　lete　results　on　p　roducts　analy　sis　and　evaluation　of　the　amount　of　nap　hthalene　in

hy　drocarbon　p　roduct　and　coke　dep　osited　on　the　cataly　st　by　using　intemal　standard

method．　From　the　view　point　ofpotential　industrial　appllca丈ion，　the　na加ral　gas　is　the

mixt　ures　of　methane　with　some　h　igher　hy　drocarbons　s　uch　as　C2　hy　drocarbons．　The

rat　e　of　aromat　ios　such　as　benzene　and　nap　hthalene　for　aromat　izat　ion　of　p　ure　methane　is

not　so　high　to　obtaine　a　commercialized　p　rocess　for　this　reaction．　lt　is　imp　ortant　to

enhance　the　rate　of　aromatic　p　roducts　by　adding　C2　hy　drocarbons　to　methane　feed．　in

view　of　aromatization　of　C2　hy　dro（mbons，　mainly　Zn／HZSM。5　and　Ga／HZ　SM－5　type

cataly　st　has　been　rep　orted．　Little　attent　ion　has　been　p　aid　to　aromatizat　ion　of　C2　on

Mo／HLZSM－5　type　cataly　st．　ln　addition，　lt　has　been　accepted　that　C2　species　is　the

intermedjates　for　benzene　forrnation　on　the　cataly　st．　lnvestignion　of　the　role　of　C2

species　is　key　crucial　to　understand　the　mechanism　of　methane　aromatization．

Therefore，　the　study　on　methane　aromatization　with　addition　of　C2　hy　drocarbens　is　of
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significance　not　only　in　academy　but　also　in　industry．　Here，　arornatization　of　methane

with　addition　of　C2　hydrocarbons　is　presented．

2　Experimental

　　　　　Mo／HZSM－5　was　prepared　by　impregnating　NH4Z　SM－5　（SiO2／AI203＝40）　with

（NH4）6M　o70244H20　aqueous　solution，　followed　by　dry　ing　at　383　K　for　4　h　and

cal　cination　at　773　K　for　4　h．　The　rati　o　ofMo／　NH4ZSM－5　in　weight　is　about　30／o．

　　　　　Cat　aly　t　ic　t　ests　were　carried　out　with　about　500　mg　cat　aly　st　p　laced　in　a　fixed　bed

continuous－flow　straight　ty　p　e　quartz　reactor　at　a　p　ressure　of　100　kPa　（1　at　m）　and　a

methane　sp　ace　velocity　of　1440　ml／gh．　The　p　roducts　were　withdraw　p　eriodically　from

the　outlet　of　reactor　and　analy　zed　by　t　wo　on－1ine　gas　chromatograp　hy　of　Shimadazu

GC－9A　and　GC－8A　equipped　with　a　flame　ionization　det　ector　（FID）　and　a

thermoconductive　detector　（TCD）　respectively．　2．20／o　Ar　in　methane　was　employed　as

intemal　standard．　Formation　rate　of　products，　methane　conversion　and　product

selectivity　were　calculated　by　an　int　ernal　standard　analy　zing　method．　Below，　the　method

is　p　resented　briefly．　Even　if　the　total　fiow　rat　e　changpd　at　outlet　from　inlet　of　reactor

（FT．．iintiet　，FT．．iO”tlet　）．　Flow　rate　of　non－reactive　intemal　standard　gas　（F，」　doses　not

change．

　　Fst　＝FTotalCKitlct　X　XstOullet　＝＝FT．t．1inlCt　X　X，tir’1ct

Here，　X，t　refer　to　concentration　of　the　intemal　standard　gas．

Formation　rate　of　i　product　（FD　can　be　cal　culated　as，

Fi＝　FT．t．le”tlet　X　XiOUtlet　＝　F，t　×　xioutlct／×stoutlet

By　the　same　way，　methane　conversion　（C．．th．．　）　can　be　cal　culated　as，

Cmcthane　＝＝　（FmethaneinlCt　”Fmc’thaneOUflet）／　Fmethaneinlet　＝＝　1　”　FmethaneOutlCt／　Fmethaneinlet

＝1”FAr　×　Xmetha．．Outlet／XArOutlet　×　Ftneth，．，irilCt

C2　conversion　（Cc2）　is　calculated　as，　Cc2　＝（　Fc2iniet　一　Fc20utiet）／　Fc2iniet

Coke　selectivity　＝　1一　（Pro’duced　HC＋‘p’　rodUced　CO）／（converted　methane＋converted　C2）．

The　raw　materials　of　methane　and　C2　hydrocarbons　are　over　99．990／o　p　ure．
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3　Results　and　discussion

3．1　Effect　of　various　C2　hydrocarbon　addition　on　formation　rate　of　benzene　and

naphthalene

　　　Format　ion　rat　e　of　benzene　and　naphalene　for　methane　aromatizat　ion　with　and

without　addition　of　about　2．40／o　C2　（in　mole）　such　as　ethane，　ethene　and　acety　lene　on　30／o

Mo／HZ　SM　一5　was　co　mp　ared　The　res　ults　are　sho　wn　in　fi　gs　1　and　2．　Fig　1　in　di　cat　es

that　the　rate　of　benzene　formation　is　increased　by　addition　of　various　C2　hy　drocarbons

to　methane　feed．　Ethane　addition　is　most　effect　ive　to　enhance　rat　e　of　benzene

formation．　Fig　2　indicates　that　the　rat　e　of　nap　hthalene　format　ion　is　increas　ed　by

addit　ion　of　ethene　and　acety　lene．　Acety　lene　addition　is　most　effective　to　enhance　the

rate　of　naphthalene　formation．

3．2　Effect　of　C2　copcentration　in　methane　feed

　　　Format　ion耐e　of　benzene　as　a　funct　ion　of　C2　concentrat　ion　in　methane　feed　are

shown　in　fig　3．　lt　is　shown　in　fig　3　that　the　rate　of　benzene　formation　is　maximized　at

the　ethane　concentrat　ion　of　4．30／o　or　at　the　acetylenelethene　concent　ration　of　about　2．40／o

in　methane　feed．　The　rate　of　benzene　formation　is　doubled　by　addit　ion　of　4．30／o　ethane

and　it　is　increased　by　500／o　by　addit　ion　of　2．40／o　ethene　or　acety　lene　in　methane　feed．

The　Rate　of　benzene　formation　contributed　from　methane　aromatization　with　or　without

addition　of　ethane　and　the　rate　of　benzene　format　ion　contribut　ed　from　ethane

aromatizat　ion　are　comp　ared　in　fig　4．　Fig　4　indicat　es　that　more　methane　and　ethane

convert　ed　t　o　benzene　when　about　4．30／o　ethane　is　added　to　methane　feed．　lt　is　also

shown　in　fig　4　that　the　rate　of　benzene　formation　is　very　high　and　quite　stable　with　time

on　stream　for　methane　aromatization　with　addition　of4．30／o　ethane．

　　　　Forrnat　ion　rate　of　nap　hthalene　as　a　function　of　C2　concentrat　ion　in　methane　feed　are

shown　in　fig　5．　lt　is　shown　in　fig　s　that　the　rate　of　naphthalene　formation　increases

sharp　ly　with　increase　of　acety　lene　concentrat　ion　in　methane　feed．　The　rate　of
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nap　hthalene　formation　increases　by　about　4　times　by　addition　of　4．30／o　acety　lene　and　it

increases　only　a　little　by　addition　of　ethane　or　ethene　in　methane　feed．　The　Rat　e　of

nap　hthalene　formation　contributed　from　methane　aromatization　with　or　without　addition

of　4．30／o　acety　lene　and　the　rat　e　of　nap　hthalene　format　ion　contributed　from　acety　lene

aromatization　are　comp　ared　in　fig　6．　Fig　6　indicates　that　much　more　methane　and

acety　lene　converted　to　naphthalene　when　about　4．30／o　acetylene　is　added　to　methane　feed．

It　is　also　shown　in　fig　6　that　the　rate　of　nap　hthalene　formation　is　very　high　and　quit　e

stable　with　time　on　stream　for　methane　aromatization　with　addition　of　4．30／o　acety　l　ene．

Nap　hthalene　formation　rat　e　for　aromatizat　ion　of　methane　with　addit　ion　of　about　4．30／o

acety　lene　over　Mo／CHZ　SM－5　at　973　K　is　up　to　80　nmolels／g　which　is　much　higher　than

the　sum　of　naphthalene　formation　rates◎ont伽ted丘om閃e　m就hane㎝d行om　4。3％

acetylene　balanced　in　Ar　gas　at　the　same　reaction　temperature．

　　　Effect　of　C2　addition　in　methane　feed　on　selectivity　of　benzene　and　nap　hthalene

products　is　comp　ared　in　t　able　1．　It　is　indicated　in　t　able　1　that　enhancement　of　the

selectivity　for　p　roduction　of　benzene　and　nap　hthalene　is　dep　endent　on　the　nature　of　C2

additives．　The　order　of　C2　additives　for　enhancing　the　selectivity　of　benzene　is　as　below，

ethane＞ethene　＞acetylene．

The　order　of　C2　additives　for　enhancing　the　selectivity　of　nap　hthalene　is　as　below，

acetylene　〉　ethene　〉　ethane

3．3　Mechanism　for　the　rone　of　C2　in　enhancing　production　of　benzene　and

naphthalene

　　　The　mechanism　for　p　romoting　effect　of　C2　hy　drocarbons　on　arornatizat　ion　of

methane　is　schemed　as　fol｝ows，

According　to　the　scheme，　the　concentration　of　intermediates　such　as　［CH．］，　C2Hy］　for

producing　benzene　is　increased　by　addition　of　C2　hy　drocarbons　in　methane　feed，　which

is　responsible　for　increase　of　benzene　formation　rate．　The　unique　role　of　acety　lene　for

enhancing　rate　and　selectivity　of　nap　hthalene　is　contributed　to　its　react　ion　with　benzene
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’　一一一　M． ．一　一’x．

to　produce　naphthalene．

CH　4

CH壷

C2H6
CzH　4
！ilil11z”2

　r C2Hy

C2H2
C2H2‘

O」一，　co

Scheme　for　the　role　of　C2　addition　to　methane　in　enhancing　production

of　benzene　and　naphthalene

4　ConcRusion

　　Methane　aromatization　with　addition　of　C2　hy　drocarbons　such　as　ethane　and　ethene

and　acety　lene　on　M　o／HZ　SM－5　cat　aly　st　is　studied．　・　Formation　rat　e　and　selectivity　of

benzene　and　nap　hthalene　is　quite　dep　endent　on　addition　of　various　C2　hy　drocarbons　in

methane　feed，　Addition　of　a11　C2　hy　drocarbons　lead　to　increase　of　benzene　formation　rate

and　ethane　addit　ion　is　most　effect　t　o　enliance　the　rate　of　benzene　format　ion．　A　much

high　formation　rate　and　selectivity　of　benzene　is　achieved　by　addition　of　about　4．30／o

ethane　in　methane　feed．　The　formation　rate　and　selectivity　of　nap　hthalene　is　most

eflrectively　enhanced　by　adclition　of　4．30／o　acety　lene．　Although　addition　of　all　C2

hy　drocarbons　lead　to　increase　of　the　active　intermediates　for　benzene　format　ion，　the

secondary　reaction　of　benzene　to　p　roduce　naphthalene　is　assisted　by　addition　of　only

unsaturated　C2　hydrocarbons，　especially　addition　of　acetylene．
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Fig　1．　Effect　of　addition　of　2．49e　C2　hydrocarbons　to　methane

　on　bennene　production　productionon　Mo／HZSM－5　at　973　K
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Fig　2．　Effect　of　addition　of　2．4％　C2　hydrocarbons　to　methane

　　　on．　naphthalene　production　over　Mo／HZSM－5　at　973　K
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Fig　3．　Effect　of　C2　concentration　in　methane　feed　on　benzene　formation　rate
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Fig　5．　Effect　of　C，　concentration　in　methane

　　　　　feed　on　naphthalene　formation　rate
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Fig　6．　Comparation　of　naphthalene　formation　rate　with　time　on　stream
for　methane　arornatization　of　with　or　without　addition　of　acetylene
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Table　1．　Eflrect　of　C2　addition　on　s　electivity　of　benzene　and　naphthalene　for

methane　aromatization　on　30／o　Mo／HZSM－5

Feed Conversion　（o／o） Selectivity　in　hydrocarbon　p　roducts（O／o）

Benzene　naphthalene　naph．／ben．

CH4 11．5 69．5 19．1 O．27

CH4
C2H6　（4．30／o）

11．8

99．9

75．3 19．4 O．26

CH4
C2H4　（2．40／o）

11．4

100

67．5 28．6 O．42

CH4
C2H2　（4．30／o）

11．2

100

58．4 35．6 O．61

Reaction　temperature：　973　K

methane　SV：1440　m1／h／9－cat

reaction　pressure：　l　atm，
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Chapter　6

CataRytic　Dehydroaromatization　of　Methane　with　CO／CO2　tovyards

　　　　Benzene　and　Naphthalene　on　Bimetallic　Mo／Zeolite　Catalysts：

　　　　　　　　　　　　Bifunctional　Catalysis　and　Dynamic　Mechanism

1．　INTRODUCTION

　　In　recent　years　a　great　challenge　and　i　ntri　g　uing　probl　em　in　heterogeneous　catal　ysis　is　the

catal　yti　c　dehydrocondensation　of　natural　gas　i　nto　useful　peterochemical　feedstocks　such　as

ethylene　by　the　oxidative　coupling　of　methane　over　Lil［MgO　and　Sm2Cig　catalysts［1］　and　the

lower　hydrocarbons　by　the　two－step　conversion　of　methane　on　Co　and　Ru　catal　ysts［2］，　while

they　have　not　been　feasi　ble　yet　due　to　the　l　ower　yields　and　selectivities．　Since　1993，　the　non－

oxidative　conversion　of　methane　to　benzene　has　been　reported［3－8］　on　Mo／HZS　M－5　catal　ysts

in　conj　unction　with　thei　r　reacti　vities，　reaction　mechanism　and　catal　yst　characterization．

Recently，　we　have　preliminarl　y　reported　that　the　Fe／Co　modified　Mo／HZS　M－5　catal　ysts　exhibit

the　higher　conversion　of　methane　to　aromatics　such　as　benzene　and　naphthalene　with　a　large

yield　of　hydrogen［9］，and　the　COICO2　additi　on　to　methane　effecti　vely　i　mprove　their　catal　yst

stabi　l　ity［10］．　Thi　s　paper　will　present　the　the　promotion　role　of　COICO2　addition　to　methane

feed　to　effectivel　y　reduce　the　coke　formation　and　i　mprove　their　catal　yst　stability　on　Mo／HSM－5

and　Fe／Co　modifi　ed　Mo／HZSM－5．　The　remarkabl　y　hi　gher　rates　and　selectivities　of　benzene

formation　i　n　the　catal　ytic　dehydrocondensation　of　methane　were　obtai　ned　by　i　ncreasing　the

pressure　（1－5　atm）　and　flow　rates　of　methane　at　973　K　on　Mo／HZSM－5　catal　ysts．　l　n　addition，

we　will　di　scuss　on　the　bifunctional　catalysi　s　of　Mo　carbide　and　HZSM－5　support，　which　have

been　characteri　zed　by　XA　FS，　TGIDTA／Mass　and　I　R　spectroscopy．

2．1　Materials　and　catalysts　preparation

　　　36wt％Mo　10ading　Mo／HZS　M－5　and　Fe／Co　modified　Mo／HZS　M－5　catalysts　were

prepared　by　impregriation（incipient　wetness）　of　NH4ZSM一一5（SiO2／A1203＝20一“1900；　surface

area，780－925　m2／g；　Toso　Co．　and　CRI　Zeolyf　t，　lnc．）　with　（NH4）6）Vlo7Cb44H20，　Co（N（）3）2　or

Fe（NO3）3　aqueous　solutions．　The　resulting　materials　are　dried　at　393K　and　calcined　at　773K

for　6　h，　similarl　y　as　reported　in　the　previous　l　i　terature［9，10］．　Powdered　Mo　carbide（Mo2C；

100－500　mm　size）　was　purchased　from　Japan　New　Meta1　Co．　and　used　as　it．　The　hybrid

catal　ysts　of　3　wt％　Mo／Si（）2＋HZSM－5　and　Mo2C＋HZS　M－5　were　prepared　by　mechanical　ly

grinding　both　components　and　by　the　coimpregnation　from　the　slurry　sol　ution．
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2．2．　Catalytic　measurement　and　kinetic　analysis

　　　The　catalytic　tests　were　carried　out　under　1－5　atmospheric　pressure　of　methane　i　n　a　continu－

ous　flow　system　with　a　quartz　reactor　of　8mm　i．d．　which　is　charged　with　O．15－O．30　g　of

catal　yst　pellets　of　20－42　mesh．　The　feed　gas　mixture　of　98％CH4（99．9qo　puri　ty）　and　2％Ar　as

internal　standard　for　anal　ysis　was　introduced　into　the　reactor　at　7．5－20　ml／min［space　velocity＝

1000－20（］×）O　ml（g－cat．h）’1］　through　a　mass　fl　ow　control　ler（Brooks　5850E）　after　flashing　with

He　at　973K　for　40min．　Hydrocarbon　products　including　C2－C4　alkanes（and／or　alkenes）　and

condensable　C6－C12　aromatics　such　as　benzene，　tol　uene，　xylene　and　naphthal　ene　deri　vati　ves

were　analyzed　by　the　on－line　FI　D　gas　chromatography　using　a　six－way　sampl　ing　valve　heating

at　533K　on　a　Porapak－P　col　umn　and　the　one－l　ine　TCD　gc　for　H2，　Ar，　CO，　CH4　and　CO2　on　an

activated　carbon　col　umn．　T　he　condensable　material　s　such　as　benzene，　tol　uene，　xylene，

naphthal　ene，　methylnaphthalene　and　anthracene　whi　ch　were　identified　by　using　the　on－line　oc－

MS（Perkin－Elmer，　A　uto　System　GC　with　910　Q－Mass）．　Using　an　internal　standard　analyzing

method，　methane　conversion，　selecti　viti　es　of　hydrocarbon　products　and　coke　forrnation　formed

on　the　catal　ysts　were　eval　uated　aocording　to　the　mass　balance　for　carbon　and　hydrogen［9，10］．

3．　RESULT　AND　DISCUSSION

3．1　Methane　aromatization　on　Mo／HZSM－5　and　Fe／Co　modified　Mo／HZSM－5

　The　non－oxidative　conversion　of　methane　was　conducted　by　flowi　ng　methane（1－5　atm）　on　the

catal　ysts　i　n　the　fi　xed　bed　at　973K　Using’2％Ar　as　the　i　ntemal　standard　for　the　reaction

anal　ysis，　the　yields　and　selectivities　of　benzene　and　coke　formati　on　were　carefully　eval　uated

according　to　the　previous　work［9，10］．　C2－hydrocarbons　and　aromatic　products　such　as

benzene　and　naphthalene　were　continuously　obtained　with　a　considerable　evoluti　on　of　hydrogen

（H2／　benzene＝9－18　mol／mol）　on　the　Mo／HZS　M－5，　Fe／Co　modifi　ed　Mo／HZSM－5　catal　ysts．　The

conversion　of　methanetca　12　90）　usual　l　y　decreases　rapi　dly　at　the　initial　stage　and　moderately　at

the　later　stage　of　the　reaction　down　to　49e　i　n　30　h　possi　bly　due　to　the　coke　formation．　The　major

hydrocarbons　produced　on　Mo／HZSM－5　consist　of　benzene（50－68％），　naphthal　ene（4512qo）

and　toluene　（2－5　％）　as　the　aromatics，　and　C2　products（412％）　such　as　ethylene　and　ethane．

The　other　carbon－containing　products　such　as　CO，　propene，　butene，　and　methylnaphthalene，

phenanthrene，　anthracene　and　tetracene　as　pOly－oondenced　aromatics　were　produced　in　a　small

amount　（totally　less　than　several　％　selectivity）．　Further　bulkry　substi　tuted　aromatic　compounds

are　minorly　obtained　on　the　catalysts　at　973K　because　of　the　higher　activity　for　hydrogenolysis

and　molecular　shape　selecti　vi　ty　due　to　the　limited　size　of　micropores　of　HZSM－5（5－6A）

aocessible　only　to　benzene　and　tol　uene．　Our　obtai　ned　results　on　the　Mo／HZSM－5　catal　yst
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show　that　the　benzene　selectiviti　es　based　on　the　consumed　methane　did　not　exoeed　42qo　at　the

maxi　m　um　val　ue　owing　to　the　formation　of　coke（32－42qo　sel）　and　polycondensed　by－products．

　　　From　Fig．1aand　lb　it　is　clear　that　the　addition　of　Fe　as　fbr　Co，ぬMo／HZSM－5　catalysts

yields　a　considerable　enhancement　of　aromatic　products　such　as　benzene　and　naphthalene　i　n　the

reaction　at　973K．　A　maxi　mum　rates（120－150　nmol／blg－cat）　and　selecti　vi　ty（52・一54qo　sel）　of

benzene　formati　on　and　minimizi　ng　coke　formation（less　than　20％）　was　attai　ned　by　the　optimum

addi　tion　of　Co（and／or　Fe）　to　MolHZSM－5．　Further　addi　ti　on　of　Fe　or　Co　to　Mo／HZS　M－5　resul　ts

in　a　decrease　of　the　aromatic　products　owing　to　substanti　al　i　ncrease　of　coke　formation　i　n　the

methane　conversion．　The　characteri　zation　by　XRD，　TGIDTA／lvlass　and　EXAFS　suggest　that

the　binary　FeMo　and　CoMo　carbides　are　hi　ghly　dispersed　i　n　the　micro－channels　of　HZSM－5

zeolite　in　the　methane　conversion　at　973K，　implying　that　Fe　and　Co　act　as　anchoring　elements

to　promote　the　dispersion　of　Mo　carbide　as　the　acti　ve　species　for　the　reaction．
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2

Fi　g．1．　Catal　ytic　performances　of　（a）　Co－Mo／／HZSM－5　and　（b）　Fe－Mo／HZSM－5　for　melh．ane

dehydroaroniatizati　on　at　973K　and　1　atm　as　the　function　of　the　Co／Mo　and　Fe／Mo　com　positi　on．

yields　a　considerable　enhancement　of　aromatic　prod　ucts　such　as　benzene　and　naphthalene　i　n　the

reaction　at　973K．　A　maximum　rates（120－1so　nmol／h／g－cat）　and　selecti　vi　ty（52－5490　sel）　of

benzene　formation　and　minimizing　coke　formation（less　than　209to）　was　attained　by　the　optimum

addi　tion　of　Co（and／er　Fe）　to　Mo／HZSM－5．　Further　addition　of　Fe　or　Co　to　Mo／HZSM－5　results

in　a　decrease　of　the　aromati　c　prOductS　owing　to　subs　tanti　al　i　ncrease　of　coke　formation　i　n山e

methane　conversion．　The　characterization　by　XRD，　T　GIDTA／Mass　and　EXLA　FS　suggest　that

the　binary　FeMo　and　CoMo　carbides　are　hi　ghly　dispersed　in血e　micro－channels　of　HZSM－5

zeolite　in　the　methane　conversion　at　973K，　implying　that　Fe　and　Co　act　as　anchoring　elements

59

摯・．

i｝一一／

、豊，∵
鯨欝轄隙il≒
　1唱

・・」

D．

DM！，ゴ泌　ら．　．・・， ミ’

鼬

t5，μ勇L



to　promote　the　dispersion　of　Mo　carbide　as　the　acti　ve　species　for　the　reac　tion．

3．2．　Effects　of　pressure　and　flow　rate

　　Fi　g．2　shows　the　dependency　of　rates　of　products　and　coke　formation　i　n　the　methane　reaction

on　3　qoMo／HZSM－5　at　973K　by　varing　the　methane　pressure　of　1－5　atm，　respectively．　I　t　i　s

intersti　ng　to　fi　nd　that　the　formati　on　rates　of　benzene　and　C2　hydrocarbons　i　ncrease　moderately

by　increasing　the　methane　pressure．　The　maxi　mum　yields　and　selectivity　（65％　sel）　of　benzene

were　attai　ned　at　34　atm　of　methane　by　the　constant　fl　ow　rate　of　2500　（ml／h／g－cat）　at　973　K，

舳ough血e　methane　conversion　is　thermOdynamically　unfavoIable　for　the　pressurized

conditi　on（6（）H4＝　C6H6＋9H2）．　Moreover，　i　t　i　s　of　worthy　to　note　that　the　coke　formation　i　s

substantially　suppressed　and　the　catalyst　stability　was　relatively　improved　by　increasing　the

methane　pressure　above　3　atm　on　the　catalyst，　as　shovvn　Fi　g．2．　This　may　be　related　with　the

sufficient　suppl　y　of　h　ydrogen　from　methane　and　the　concentration　of　the　surface　carbon　species

CHx　for　the　formati　on　of　arom　ati　c　products　such　as　benzene　and　naphthal　ene　on　the　catal　ysts．

　　On　the　other　hand，　by　i　ncreasi　ng　the　flow　rates　of　methane　refered　to　as　the　speci　fi　c　veloci　ty

（mi　th／g－cat）　from　1000　to　2rmO　the　rate　of　benzene　formati　on（nmol／s／g－cat）　under　the　3

atmosphere　of　methane　at　973K　on　3％Mo／　HZSM－5　i　s　remarkabl　y　enhanced　by　ca．　10　times　as

shown　in　Fi　g．　3．　The　methane　conversion　at　the　5　h　time－on　stream　moderately　decreased

from　8　to　6　90　by　changing　the　speci　fi　c　velocity　from　1000　to　20000（ml／h／g－cat）．　These　resuks

suggest　that　benzene　i　s　produced　under　the　diffusion　con　trol　on　the　catal　ysts　and　the　methane

conversi　on　to　benzene　is　close　to　the　thermodynamical　equi　libri　um（benzene　conversion　for　H2／
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Fi　g．3．　Product　formation　rates　and　methane　conversion　on　3　qowtMo／HZSM－5　by　vary－

ing　specific　velocity　of　methane（ml／h／g－cat）　at　3　atm　and　973K
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C6H6＝12　at　973K　is　calculated　as　8．2％　and　1　atm　of　methane）　under　the　reaction　conditions．

The　effecti　ve　transportation　of　the　aromati　c　products　resuks　i　n　the　hi　gher　conversion　of　methane

and　suppressing　the　coke　formation　through　the　micro－channels　of　HZSM－5．

3．3．　Selective　dehydroaromatization　of　methane　by　CO／CO2　addition

　The　addition　of　CO　to　methane　feed　gas　exerts　a　significant　effect　on　the　catalyti　c

performances　of　Mo／HZS　M－5　and　Fe／Co　modifi　ed　Mo／HZS　M－5　for　the　aromatization　reacti　on

to　promote　the　benzene　formation　and　catal　yst　stability．　Using　pure　CH4　as　the　feed　gas，

methane　conversion　and　benzene　forrnati　on　rate　were　decreased　greatly　duri　ng　2・4　hours　of

reaction　time　on　stream，　as　shown　i　n　Fig．　4．　While　with　the　addition　of　CO　in　the　feed　gas，　the

methane　conversion　drop　was　substantially　moderated　regardl　ess　of　varying　the　partial　pressure

of　CO　（from　1．790　to　12．3　qo）．　A　fter　attaining　a　i　ncrease　of　benzene　formation　in　the　earl　y　stage

of　the　meth～me　conversion　with　CO，　benzene　formation　rate　kept　almost　constant　du血g出e

prolonged　reacti　on　up　to　more　than　24　hours．　l　t　i　s　worthy　to　note　that　in　the　steady　reaction

stage，　the　amount　of　CO　is　al　most　ke　pt　constant　before　and　after　the　reaction　i　n　al　1　the　case　of

CO　addition．　The　results　imply　that　CO　is　regenerated　possibl　y　due　to　the　reaction　of　the　surface

61

＃．su

し

騨



carbon　wi　th　the　active　oxygen　species　deri　ved　from　CO　dissociation　as　shown　i　n　Scheme　1．

　　By　addi　tion　of　CO　to　a　1　atm　methane　fl　ow　at　973K　the　Co　and　Fe　modified　Mo／HZS　M－5

catalysts　at　973K　and　1　atm　of　methane　exhi　bit　remarkable　catal　ytic　activities　for　the　benzene

formation　in　high　selectivities　（75－859（o　sel．　at　10－13‘7e　CH4　conv．），　as　shown　i　n　Fi　g．　4．　The

rate　of　benzene　formation　decreases　moderatel　y　but　keep　the　hi　gher　rates　and　l　ower　coke

selectivity（less　than　10％）　for　100　h．　Similarly，　the　stability　of　the　catalyst　was　also　improved

by　adding　a　few　pe！　cent　of　CO2　into　the　feed　gas．　The　addition　of　1．6ケ日CQ2　yielded　a　higher

methane　conversion　and　hydrogen　formation　rate．　Nevenheless，　the　benzene　formation　was

greati　y　suppressed　by　the　excoeding　addition　of　CO2（more　than　12qo），　but　i　t　yields　the

improvement　of　the　catal　yst　stability　i　n　the　methane　aromatization　duri　ng　the　prol　on　ged　ti　me．　I　t

is　suggested　that　CO2　i　s　converted　with　CH4　to　produce　double　amount　of　CO　and　H2　under　the

reaction　condition　by　the　reforming　process　（CO2＋CH4＝2C（）＋2H2）．　The　formed　CO　may　exert

asimilar　promotion　of　pr（）ductS　formation　and血e　catalytic　stability．　The　excess　CO2　may

greatly　consume　the　surface　active　carbon　speeies　such　as　CHx　and　C2　species　for　the　reaction．
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Fig．　4．　Rates　of　benzene　formation　in　methane　aromatizati　on　with　and　wi　thout　addition　of　CO

（1　atm，　3．5％CO，　973K）　on　Mo／HZSM－5　and　Co－Mo／HZSM－5　catalysts．
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Fig．5．　TPO　pattterns　as　CO2　form－a－t．ion一．　by　floyving　a．i．’r－Ao．nv　the．ry！g！HiZI．SA！yt－5　catalysts　after　the

rea6tion　withi　CH4，　CH4＋CO　and　CH4＋C（）2　at　1　atm，　973K　and　SV＝1500
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　　As　shown　in　Fi　g．5，　it　was　demonstrated　by　the　temperature－programmed　oxidation（TPO）

e脚mentS（㎜ping舳即ture　rate：2K／mi　n，　ai　r　flow　of　20　ml／min）that　the　amount　of　coke

deposi　tion　on　the　catal　yst　surface　was　greatl　y　reduced　by　adding　vari　ous　amount　of　CO　or　CO2

to　the　methane　feed　gas．　The　increase　of　CO　concentration　from　1．7％　to　12．0％　i　n　methane

flow　resulted　in　a　marked　suppression　of　the　total　coke　formation　on　the　catalyst　surface，

particularly　the　irreversible　or　i　nert　coke　w　hich　was　oxidized　to　CO2　at　the　higher　temperature

above　773K　in　TPO　experi　ment．　The　i　ncrease　of　CCL．）．　partial　pressure（1．4　to　4．Oqo）　in　methane

feed　gas　decreased　effectively　the　coke　forrnation　on　the　catalyst　to　a　much　lower　level，

compared　with　that　of　CO．　This　may　be　related　to　i　ts　hi　gher　activity　of　the　dissociated　oxygen

in　the　reaction　with　su血ce　carbOn　to　form　CO（CO2＋C一一一＞2CO）as　shown　in　scheme　1・

To　understand　the　role　of　CO　in　methane　aromatization　reaction，　the　i3CO＋CH4　reaction　was

conducted　in　a　closed　circulating　reaction　system．　The　starting　composition　of　the　mixture

reactant　is　8qo　i3C（）＋9290　CH4　with　a　total　pressure　around　220　torr．　Even　after　5min　of　the

closed　circulating　reaction　at　973K，　i3C　atoms　involved　in　methane　（i3CH4）　reached　the　value　of

8％．　The　isotopic　abundance　of　i3C　incorporated　in　benzene　molecule　were　detected　by　GC－

Mass　analysis　as　the　random　distribution　（i3CCsH6－47．6qo，　i3C2C4H6－36．7qo，　i3C3C3H6－

13．0％，　i3C4C2H6－2．4qo，　i3CsCH6－O．3％）　regardless　of　the　reaction　ti　me　of　methane　on
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Mo／HZSM－5．　These　results　suggest　that　the　carbon　derived　from　CO　dissociation　is　efficiently

incorporated　into　the　benzene　formation　and　the　isotopic　scrambling　reaction　between　i3cO　and

i2CH4　proceeds　rapidl　y　under　the　reaction　condition　at　973K．　B　ased　on　the　above　results　we

suggest　the　mechanism　of　an　unique　role　of　CO　in　methane　aromati　zation　reaction　to　stabilize

the　caぬl　yst　performances　and　promote　the　benzene　formation　as　shown　in　scheme　1．　Firstly，

CO　dissociates　on　the　Mo　sites　to　form　the　active　carbon　species　CHx　and　C2　species　through

reaction　（2）　and　（3）；　followed　wi　th　their　ol　i　gomeri　zation　to　form　hi　gher　hydrocarbons　such　as

benzene　and　naphthalene　in　reaction　（1）　on　the　catalyst；　The　dissociated　oxygen　speci　es　｛O］

from　CO　may　react　with　the　s　urface　i　nert　carbon　species　（coke）　to　regenerate　C　O，　resul　ting　i　n

the　suppression　of　the　coke　formati　on　on　the　catal　yst　throu　gh　reaction　（4）．

Table　1．　Catal　yti　c　performanoes　of　the　hybridcatalysts　of　MoC2（or　Mo（）3／SiO2）＋

HZSM－5　in　the　methane　aromati　zation　at　1　atm　and　973K

　　　　　　　　　　　catalyst　CH4　conv（　（70）　V（Benzene）a

HZSM－5

Mo2C

MoC7＋HZSM－5
　　　　一

MoO3／SiO2

MoO3／SiO2＋HZSM－5

3qoMo／HzsM－s

1．2

08

5．7

5．3

6．2

8．5

O．6

3

150

412

2019

3162

a　Specific　rates　of　benzene　forrnation（nmol／Mo／sec）

3．4．　Bifunctional　catalysis　by　Mo　carbide　and　ZSM－5　zeolite　support

　　EXAFS（Mo　K－edge）　and　TGIDTAIMass　studies　have　been　performed　in　methane

conversion（1　atm，　300－973K）　on　6wt％　Mo／HZSM－5　using　the　SOR　facility（KEP－10　line）　and

Mac　Science　system（TG－DTA2（）20S），　respectively．　I　t　was　demonstrated　that　Mo　oxide　on

HZSM－5　i　s　converted　with　methane　at　923K　by　the　evol　ution　of　H2　and　CO　to　Mo　carbide

cl　ustertMo2C；　Mo－Mo；　C．N．＝　2．1－3．5，　R－2．98A），　where　the　methane　aromatization　reaction

initiats　to　form　C6H6，　C　loHs　and　C2H4　in　evolving　hydrogen　above　923K．　The　results　imply

that　Mo　carbide　on　HZS　M－5　support　i　s　an　acti　ve　species　for　the　catal　ytic　methane　aromatizati　on

similarly　as　it　has　been　proposed　peviously　by　Solymosi　and　Lunsford［5，6，8］．

　　To　understand　the　bifuncti　onal　role　of　Mo　carbide　and　HZSM－5　support　the　following
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experiments　were　conducted．　As　indicated　in　Table　1，　it　was　of　interest　to　find　that　the　specific

rates　of　benzene　formation（C6HelMo／sec）　on　a　hybrid　catalysts　of　MoC2（or　Mo（）3／Si　O2）

＋HZSM－5　at　973　K　were　greatl　y　enhanced（50－100　ti　mes）　by　the　mechani　cal　mixture　of　each

component，　which　shows　neglegible　or　low　activity．　These　imply　the　bifunctional　catalysis　for

methane　aromatizati　on　at　the　i　nterface　of　Mo　carbicle　and　ZS　M－5　support．　T　he　resul　ts　by　i　n－situ

FTI　R　and　EXA　FS　studies　suggest　that　methane　i　s　acti　vated　on　Mo　carbide　sites　to　form　CHx

and　C2－species（e．g．CH3－C；2967，　2878，1468　and　1382　cm－1）　which　may　migrate　at　the

interface　of　ZSM－5　acidi　c　support，　followed　by　the　dehydr（）aromatization　of　C2－intermedi　atesぬ

benzene　and　naphthalene．　A　s　it　i　s　discussed　in　the　session　3．3．，　the　common　carbon　species

CHx　by　the　dissociation　of　CO　and　methane　on　MoC2　are　converted　to　C2一一hydrocarbons，

benzene　and　naphthal　ene　on　the　acidic　sites　of　the　ZS　M－5　support．　T　he　infl　uence　of　Si（）2／A12Cig

rati　o（20－1900）　of　the　HZSM－5　zeolite　on　benzene　prod　uction　and　coke　formation　were　al　so

investi　gated．　l　t　was　found　that　there　exi　sts　a　shaip　optimum　SiQ2／Ai203　ratio　of　40　yielding

hi　ghest　rates　of　benzene　fomiati　on　and　lowest　coke　formation．　The　HZSM－5　supports　havi　ng

too　smaller　or　1arger　S　i／Al　ratios　beyond　40　produoe　the　Mo／HZS　M－5　catalysts　of　poor　acti　vity

for　methane　aromatization．　T　he　proper　B　ronsted　acidi　ty　of　HZS　M－5　promotes　the　conversion

of　C2－intermediates　to　benzene　and　naphthalene　and　suppress　the　coke　formation．　T　hi　s

bi　functi　Dnal　catal　ysi　s　of　Mo2CIHZS　M－5　for　methane　aromati　zation　reacti　on　is　discussed　by

analogy　with　the　synergeti　c　promotion　on　Pt／A12CB　catal　yst　for　the　dehydroaromati　zation　of

alkanes．

4．　Conclusion

（1）　Catal　ytic　dehydroarom　ati　zation　of　methane　with　CO　and／or　CO2　proceeds　at　｛；73　K　under　the

higher　pressure　of　methane（1－5　atpa）　on　Mo／HZSM－5　and　Fe／Co　modified　Mo／HZSM－5

towards　aromatic　products　such　as　benzene　and　naphthalene　with　hi　gher　yiel　ds　and　stable

selectivities（above　85％）and　minimizing　the　coke　formation　in血e　prolonged　time　of　100　h．

（2）The　catalyst　stability　by　adding　CO　and／or　CO2　to　the　methane　feed　is　ass㏄iated　with

effecti　vely　suppressing　the　coke　formation　due　to　CO　dissociation　on　Mo／HZS　M－5　catal　ysts．

（3）　A　remarkable　synergeti　c　promoti　on　for　benzene　formati　on　i　s　obtai　ned　on　the　hybrid　catal　ysts

consisting　of　Mo／SiO2＋HZSM－5　and　MocC＋HZSM－5，　whereas　each　component　shows　a

marginal　activity　in　the　reaction．

（4）　Bifunctional　catal　ysis　of　Mo／HZSM－5　i　s　di　scussed　by　the　mechanism　that　methane　is

acti　vated　on　Mo　sites　to　form　CH：　and　C2－species，　which　may　migrate　at　i　nterfaoe　of　ZSM－5

having　the　optimum　acidity　for　the　dehydrocyclization　to　benzene　and　naphthalene．
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Chapter　7
1nhibitation　of　coke　deposition　and　activity　enhancement　for　methane

aromatization　by　varying　reaction　conditions　on　Mo／HZSM－5

Abstract：

　　　Dep　endence　of　activity　and　selectivity　for　methane　aromatization　on　M　onz　SM－5

0n　the　reaction　condit　ions　s　uch　as　p　res　s　ure　and　sp　ecific　velocity　of　methane，　C　O2

addition　at　high　SV　and　reaction　temp　erature　is血vestignted・　A　high　and　stable　benzene

formation耐e　of　about　1700　nmole／s／g－cat　is　achieved　for　methane　aro鰍lzatlon　on

Mo／HZ　SM－5　by　increasing　methane　pressure　from　1　to　3　atm，　methane　specific　velocity

from　1440　to　1000（）　ml／h／g－cat　and　reaction　temperature　from　973　to　I　Cr73　K，　and　by

adding　about　2．　lqo　CO，　to　methane　feed．　The　rate　of　benzene　formation　with　time　on

stream　is　remarkably　stabilized　by　v　arying　methane　pres　s　ure　and　addition　of　2．　1％　CO，

in　methane　feed　・nd　greatly　enh・nced　by　inc・・紬9　methane　specific　vel・ciry　f・・m　14中O

to　10000　ml／h／g一一cat　and　reaction　temperature　from　973　to　1073　K．　The　selectivity　of

benzene　and　naphthalene　and◎oke　is　also　oontrolled　by　vary血g　reaction　conditions・

High　methane　specific　velocity　lead　to　high　benzene　selectivity　and　low　naphthalene

selectivity．　Coke　selectivity　is　minimized　at　3　atm　methane．
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1　lntroduction

　　　In　recent　y　ears，　a　great　challengo　and　intriguing　p　roblem　in　het　erogeneous　cat　aly　sis

is　the　catalytic　dehy　drop　oly　merization　of　natural　gas　into　useful　p　etrochemical

feedstocks　and　liquid　fuel．　M　ethane　aromatization　on　Mo／HZ　SM－5－based　cataly　sts

has　been　attract　ed　a　in　creasing　attention．　The　convent　ional　reaction　condit　ions

employ　ed　for　methane　arornatization　on　Mo／HZSM－5　are　at　methane　pressure　of　about

1－2・atm，　methane　specific　vel・city・f　1440㎡／・／gcat　and　t・mpe耐u・e・f973　K［1－20］・

In　the　present　study，　the　role　of　reection　conditions　such　as　pressure　and　specific

velocity　of　methane，　CO2　addition　in　methane　feed，　reaction　temp　erature　and　Mo－

loading　on　HZ　SM－5　sup　p　ort　in　controlling　the　activity　and　select　ivity　for　methane

aromatization　is　discussed．
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2　Experimental

　　　　　Mo／HZSM－5　was　prepared　by　impregnating　NH4ZSM－5　（SiO2／AI203＝40，　Toso

Co．）　with　（NH4）6Mo70244H20（Kanto　Chem　Co．）aqueous　solution，　followed　by　drying

at　383　K　for　4　h　and　calcination　at　773　K　for　4　h．　The　samp　le　was　fmally　p　ressed，

crus　sed　and　sort　ed　into　20－40　mesh．　Cataly　tic　t　ests　were　carried　out　with　about　500

mg　cat　aly　st　p　laced　in　a　fixed　bed　cont　inuous－flow　straight　ty　p　e　quart　z　react　or．　Aft　er

cat　aly　st　was　p　ret　reat　ed　by　air　st　ream　at　873　K　for　30　min　and　t　hen　by　He　stream　at　873

K　for　30　min，　the　stremn　was　s　witched　to　feed　gas　an　d　cat　aly　st　bed　was　brought　to

，eacti。n　t・・叩e伽e．　The　p・・duCts　w・・e　Withd・aw　p・ri・dl・nlly　fr・m　the・utlet・f

react　or　and　analy　zed　by　two　on－line　gas　chrornatograp　hy　of　Shimadazu　GC－9A　and　GC－

8A　equip　p　ed　with　a　flame　ionizat　ion　detector　（Fil））　and　a　therrnoconductive　detector

　（TCD）　respectively．　Hydrocarbon　products　were　separated　by　a　4mm　×2m　Porapak　P

colurnn　and　Ar　，　CH4，　CO，　CO2，　H2　were　analy　zed　by　a　4mmx　l　m　active　carbon　column．

2．20／o　Ar　in　methane　was　emp　loy　ed　as　intemal　standard．　Formation　rate　of　p　roducts，

methane　convers　ion　and　p　roduct　selectivity　were　calculated　by　an　internal　standard

an　alyzing　method　［　1　9］　．　Fo　rm　ation　rate　of　produ　cts，　meth　ane　conversion　and　pro　duct
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N

select　ivity　were　calculated　by　an　internal　standard　analy　zing　method．　B　elow，　the　method

is　p　resented　briefly．　Even　if　the　total　flow　rate　changad　at　outlet　from　inlet　of　reactor

（FT．ulintle’　，FT．ulO“tiet　）．　Flow　rat　e　of　non－react　ive　intemal　standard　gas　（F，D　doses　not

change．

F，t　＝FT．t，10”tlet　XX，tOUtlet　＝　FT．t．1inlet　xx，tinlet

Here，　X，t　refer　to　con　centration　of　the　internal　standard　gas．

Formation　rate　of　i　product　（FD　can　be　calculated　as，

Fi＝FT．t．iOutict　XXiO”tiet　＝F，t　×　x｛outiet／x，teutict

By　the　same　way，　methane　conversion　（C．．th．．　）　can　be　cal　cul　ated　as，

Cmethane　＝＝　（Fmethaneinlct　一FmethaneOUtlet）／　］Fmethaneinlet　＝　1一　Fmethen．Out：ct／　Fmethaneinlet

＝1’FAr　×　XmethaneOutlet／lXAxOUtlet　×　Fm．th．n．inlet

Coke　selectivity　＝　1一　（p　roduced　HC＋p　roduced　CO）／converted　methane．　The　raw

materials　of　methane　and　ethane　is　over　99．990／o　pure．

‘

ド

3　Results　and　discussion

3．1　Effect　of　methane　pressure

　　　　The　reaction　for　benzene　aromatizat　ion　was　conducted　by　vary　ing　p　ressure　of

methane．　Benzene　forrnation　rate　with　time　on　stream　at　difi7erent　methane　p　ressure　is

shown　in　fig　1．　At　the　iower　methane　p　ressure　less　than　1　atm　（including　1　atm），

benzene　format　ion　rate　decline　with　t　ime　on　st　ream　although　it　is　higher　at　the　initial

stage．　At　the　higher　p　ressure　of　2－3　atm，　benzene　format　ion　rate　increase　with　time　on

strearn　at　the　initial　stage　and　then　become　’stable．　However，　at　so　high　methane

pressure　as　5　atm，　the　formation耐e　is　somehow　suppressed　although　a　stable耐e　of

benzene　formation　is　still　available．　Therefore，　a　stable　rate　of　benzene　format　ion　is

achieved　by　conduct　the　reaction　at　the　higher　methane　p　ressure　higher　than　2　atm．　The

effect　of　methane　p　ressure　on　the　selectivity　of　coke　and　format　ion　rat　e　of　hy　drocarbon

products　is　shown　in　fig　2．　The　selectivity　of　coke　is　minimized　and　the　selectivity　and

format　ion耐e　of　benzene　is　maXimized　increasing　methane　p　ressure　from　O．28　at　m　t　o　3
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atm　（fig　2）．　Coke　selectivity　at　3　atm　methane　is　about　10　times　less　than　that　at　O．28

atm　methane．　High　methane　pressure　lead　to　low　coke　selectivity，　which　is　responsible

飴r助stめility鉤r　benzene飴舩i・n・F・舩i・n鵬・f　ethene　and　ethane　is

also　quite　dependent　on　the　pressure　ofmethane　as　shown　in　fig　3．　Ethene　formation

rat　e　decreas　es　and　ethane　formation　rat　e　increas　es　with　increasing　methan　e　p　art　ial

pressure．　Higher　partial　pressure　of　methane　is　favorahle　for　transformation　of　ethene　to

ethane　through　hydrogen　transfer　reaction．　The　coke　species　deposited　on　the　catalyst　at

different　methane　partial　pressure　is　characterized　by　TPO　spectra　．　Two　distinct　coke

species　respond　to　different　oxidation　temperature　can　be　identified　by　TPO　spectra　as

shown　in　fig　4．　The　two　TPO　peaks　are　ascribed　to　the　coke　species　（Cm）

dep　osited　on　metal　sites　formed　by　deep　dehydrogenation　of　hydrocarbon

species　and　coke　species　（Cs）　deposited　on　aeid　sites　of　support．　lt　seems

that　high　methane　pressure　is　unfavorable　to　formation　of　the　coke　species

　（Cs）　deposited　on　support．　However，　the　coke　species　（Cm）　on　metal，　which　is

contributed　to　formation　of　carbide，　is　independent　of　methane　pressure．
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3．2　Effect　of　methane　specific　velocity　and　reactien　temperature

　　　　Format　ion　rat　e　of　benzene　with　time　on　stream　at　3　at　m　methane　and　different

methane　specific　velocity　is　shown　in　fig　5．　At　a　methane　specific　velocity　of　1440

ml／h／gcat，　formation　rate　of　benzene　is　about　130　nmolels／g－cat　and　quite　stable　with

time　on　stream．　B　enzene　format　ion　rat　e　increas　es　remarkably　with　increas　ing　methane

sp・ecifi・v・1・・ity．H・wev・・，　b・nzene・f…nati・n耐・輌th　time・n　st・eam　decline　at　high

methane　specific　velocity．　The　coke　species　deposited　on　the　catalyst　at　different

methane　specific　velocity　is　characterized　by　TPO　spectra　as　shown　in　fig　6．　lt

indicates　in　fig　6　that　high　methane　specific　velocity　mainly　lead　to　increase　of　the　coke

species　of　Cs．　in　addition，　product　selectivity　is　also　dependent　on　methane　specific

velocity　as　shown　in　fig　7．　Benzene　selectivity　remarkably　increases　and　naphthalene

selecti　vity　decreases　with　increasing　methane　specifi　c　velocity　from　720　to　2500　ml／h／g，
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an　d　C2　selectivity　is　also　increas　es　with　increase　of　methane　specific　velocity．　This

refleCt　some　mechanistic　fea加res　of　methane　aromatization．　Benzene　ls　pnmary

aromatic　p　roducts　and　nap　hthalene　is　p　roduced　by　the　s　econdary　reaction　of　benzene．

Therefore，　benzene　selectivity　is　enhanced　and　nap　hthalene　selectivity　is　sup　p　ressed　at

a　higher　specific　velocity　of　methane　since　high　SV　is　unfavorable　for　the　secondary

reaction．　On　the　other　side，　high　nap　hthalene　selectivity　is　available　at　low　methane

specific　velocity．　Naphthalene　selectivity　at　a　lower　methane　specific　velocity　of

about　720　rnl／h／g　is　about　400／o．

3．3．　Effect　of　reaction　temperature　arnd　Mo－loadings

　　　　Benzene　format　ion　rat　e　also　increas　es　quit　e　dramat　ically　with　increas　e　of　react　ion

temp　erat　ure　as　shown　in　fig　8．　Rat　e　of　benzene　formation　is　almost　doubled　by　50　K

increas　e　from　973　K　to　1023　K．　Therefore，　a　very　high　benzene　formation　rat　e　is

achieved　at　high　specific　velocity　and　high　reaction　t　emp　erat　ure．　At　the　severe　react　ion

conditions　of　higher　SV　and　higher　reaction　temp　erature，　benzene　formation　rate　and

coke　selectivity　is　also　dep　endent　on　the　M　o－loading　of　M　o／HZ　SM－5　cataly　st．　The

coke　selectivity　is　minimized　and　benzene　formation　rate　is　maximized　at　the　Mo－

loading　of　about　wt．6－21　O／o　as　shown　in　figure　9．

3．4　Effect　of　C　O2　addition　in　methane　feed

　　　Although　rat　e　of　benzene　format　ion　is　increased　remarkably　by　increas　ing　methane

specific　velocity　and　react　ion　temp　erat　ure，　benzene　format　ion　rate　decline　sharp　ly　with

time　on　stream　because　of　the　serious　coke　deposition　on　the　cataly　st　at　high　methane

specific　velocity．　To　stabilize　the　cataly　st　at　high　methane　specific　velocity，　about

2．10／o　CO2　is　added　to　the　methane　feed．　Fortunately，The　rate　of　benzene　formation　is

stabilized　with　time　on　strearn　at　high　methane　specific　velocity　by　addition　of　2．10／o

CO2　to　methane　feed　as　shown　in　fig　10．　lt　is　shown　in　fig　I　O　that　a　high　and　stable

benzene　fo　rmat　ion耐e　of　about　lOOO　nmole／s／g一（識is　achieved　for　methane

71

幽
．

ρ

■
」
　
一

曹
曾
比

．
」
層
砥
・
」
・
～
－
－
f
‘
乳
－
．
．
己
」
1
－

＝
8
曙
一
・
は
、
ゴ
ー
」
～
．
丼
一
．
ー
ー

『メ

E

　
覧
　
．
．
昏

葎
，
、
3
同

割
　
　
　
・
－

畷
驚



一■一嗣一■一剛『■騨■■「＝’一一一〒＝『’「］」一

鉾・

一・・

　　　　U；

aromatization　with　about　2．10／o　CO2　on　150／oM　o／HZ　SM－5　at　1023K　and　methane

specific　velocity　of　IOOOO　ml／h！gcat．　in　addition，　the　rate　of　benzene　formation　for

methane　arom加ization　with　CO2　is　greatly　increased　f㌔耐her　by　mcreasmg　reactlon

temp　erature　from　1023　K　to　1073　K．　A　very　high　benzene　formation　rat　e　of　about

1700　nmole／s／gcat　is　attained　for　methane　aromatization　with　2．10／o　CO2　on　150／o

Mo／HZ　SM－5　at　methane　sp　ecific　velocity　of　10000　ml！h／gcat　and　1073　K　as　shown　in

fig　10．　Therefore，　it　seems　that　CO2　addition　is　the　solution　to　solve　the　problem　of

the　serious　coke　dep　osition　on　the　cataly　st　for　methane　aromat　izat　ion　at　high　methane

specific　velocity　and　high　temperature．

4　Conclusion

High　and　stable　rate　of　benzene　formation　has　been　achieved　for　methane　aromatization

on　Mo／HZSM－5　by　optimizing　reaodon　conditions　including：

1）　increasing　methane　pressure　from　1　to　3　aM）．　2）　increasing　methane　specific　velocity

from　14zto　to　10000　ml／h／g－cat　and　reaction　temperatu　re　from　973　to　1073　K，　and　3）

adding　about　2．　190　CO，　to　methane　feed．　The　role　of　1）　and　3）　is　to　enhaiice　the　stability

for　benzene　formation　rate　w　ith　time　on　streani　and　the　role　of　2）　is　to　i　ncreas　e　rate　of

benzene　forrnation　rate．　The　rate　of　benzene　forrnation　is　about　1700　nmole／s／g－cat　at　the

optimized　conditions，　which　is　about　17　times　of　that　at　the　conventipnal

conditions（S　V＝1440　mlfh／g－cat，　T＝　973　K　and　P＝　l　atm）．　The　selectivity　of　benzene

and　nap　hthalene　and　coke　is　also　controlled　by　reaction　conditions．　High　methane

specific　velocity　lead　to　high　benzene　selectivity　and　low　nap　hthalene　selectivity．

Coke　selectivity　is　minimized　by　varying　methane　pressure．
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Fig　1．　Benzene　formation　rate一．with　time　Qn　stream．　for
　　　　　　　　　　　　　　　　　　　　　　　　　　　　　at　various　methanearomatization　on　3％　Mo／HZSM－5
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Fig　2．　Formation　rate　and　selectivity　of　product　as　a　function　of　paethina－n－e一一p“rcssure

　　　for　methane　aromatization　for　methane　aromatization　on　390　Mo／HZSM－5
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Fig　5． Benzene　formation　rate　with　time　on　stream　for　methane　aromatization
on　3‘）6　Mo／HZSM－5　at　various　methane　specific　veaocity
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　　　7．　Dependence　of　benzene　product．ionFig
o’氏@396　Mo／HZSM－5　at　3　atm　and　973　K

on　methane　specific　velocity
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Fig　8．　Dependence　of　benzene　formation　rate　on
methane　specific　velocity　and　reaction　temperature
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Fig　9．　Formation　rate　and　selectivity　of　benzene　as　a　function　of

Mo－loadings　of　McNtHZSM－5　catalyst　for　methane　aromatization
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Fig　10．　Effect　of　CO，　addition　in　methane　feed　on　the　stability　of　benzene

formation　for　methane　aromatization　on　15‘）6　Ma／HZSM－5
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Chapter　8
Catalytic　performance　of　Re／HZSM－5　for　dehydro一（Hme血ation　and

dehydro－aromatization　reactions　of　methane

Abs血・act

　　Cat　aly　tic　feat　ures　of　Re／HZ　SM－5－bas　ed　cat　aly　st　for　aromatiz　at　ion　react　ion　of

methane　are　studied　and　comp　ared　with　that　of　Mo圧疵SM－5．　It　is　found　in　TPR　of

Re207／HZ　SM－5　in　methane　flow　that　methane　activat　ion　p　roces　s　is　init　iat　ed　by

reduct　ion　of　Re207／HZ　SM－5，　which　is　endothermic　and　comp　any　with　p　roduct　ion　of

hy・dr・ca・b・n・Xides　and・1・ss・f　catalyst　weight・H・weve・，　the・eduCti・n・f酬yst血

hy　drogen　stream　is　a　exothemic　process　and　takes　place　at　a　much　lower　temp　era加re．

We　p　rop　ose　that　rhenium　carbide　is　formed　by　reduCt　ion　of　Re207／HZ　SM－5血methane

fiow．　The　carbide　is　stable　in　the　atmosphere　of　methane　and　cataly　ze　the　reaction　of

methane　aromatization．　The　temp　erature　for　format　ion　of　rhenium　carbide　is　about　100

K　lower　than　that　for　format　ion　of　moly　bdenum　carbide　and　Re／HZ　SM－5　shows　much

better　low－temp　erature　activity　than　Mo／HZ　SM－5－based　cataly　st．　To　comp　are　with

Mo／HZ　SM－5－based　cataly　st，　Re／HZ　SM－5－based　cataly　st　exhibits　a　better　lower－

temperature　activity　and　a　higher　selectivity　to　C2　hy　drocarbons．
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．1　lntroduction

　　　Direct　transformation　of　methane　to　more　valuable　hy　drocarbons　such　as　aromatics

or　C2　hy　drocarbons　is　fas　cinating　and　highly　desirahle　for　more　reasonable　utilizing

abundant　na加ral　gas　resources．　Recently，　much　attractive　results　fbr　dehy　drogenation

and　aromatization　of　methane　over　M　o／HZ　SM－5　cataly　st　have　been　rep　orted　and　it　has

attract　ed　a　wide　attention　and　interest　（1－20）　over　the　p　ast　several　y　ears．　lt　has　been

suggested　that　moly　bdenum　carbides　sup　p　orted　on　HZ　SM－5　is　the　active　p　hase　for

methane　aromatization．　At　the　p　resent　research，　the　catalytic　performance　of　Re／HZS　M－

5－based　cataJysts　for　dehydrogenation　and　aromatization　of　methane　is　presented　as　a

altemative　choice　of　catalyst．

2　Experimental

　　　　　Re／HZ　SM－5　was　p　rep　ared　by　imp　regnating　NH4Z　SM－5　（SiO2／AI203＝40，　Toso

Co．）　with　NH4ReO44H20　（Kanto　Chem．　Co．）　aqueous　solution，　followed　by　drying　at

383　K　for　4　h　and　calcination　at　773　K　for　4　h．　The　s　amp　le　was　fmally　p　ressed，

crushed　and　sorted　into　20－40　mesh．

　　　　Cat　aly　t　ic　t　ests　were　carried　out　with　about　500　mg　cat　aly　st　p　laced　in　a　fixed　bed

。・n伽・us－fi・w　st・aight　iyp・quartz・eaCt・・．　Th・・aw舩・rials・f　m・th・ne　i・・ve・

99．990／o　pure．　All　the　flow　rate　of　gas　was　controlled　with　mass　fiow　controllers

（ASAM　Keiki　Co．　LTD）．　The　p　roducts　were　withdraw　p　eriodically　from　the　outlet　of

react　or　and　analy　zed　by　t　wo　on－line　gas　chromatograp　hy　of　Shimadazu　oc－9A　and　（］K　一

8A　equipped　with　a　flame　ionization　detector　（FID）　and　a　therrnal　conductivity　detector

　（TCD）　respectively．　Hydrocarbon　products　were　separated　by　a　4mm　×lmPorapak　P

column　and　Ar　，　CH4，　CO，　CO2，　H2　were　analyzed　by　a　4mmx　l　m　active　carbon　column，

IO．70／o　Ar　in　methane　was　emp　loy　ed　as　internal　standard．　Format　ion　rate　of　p　roducts，

methane　conversion　and　p　roduct　selectivity　were　calculated　by　an　internal　standard

analy　zing　method　described　elsewhere　［19］．　Below，the　method　is　p　resented　briefly．
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Even　ifthe　t。tal・fl。w　rate　ch、ngOd・at・utlet　fr・m血1et・f・reaCt・r（F，．』1流uet，FT幽1．udct）・

Flow　rate　of　non－reactive　intemal　standard　gas　（Fst）　doses　not　change．

IFst　＝　FT．t．iO”tiet　XX，tOutict　＝FTot．iiniet　xxstiniet

Here，　X，t　refer　to　concentration　of　the　intemal　stan　dard　gas．

Formation　rate　of　i　product　（FD　can　be　calculated　as，

Fi　＝FT．t．iOutiet　XXiOutiet　＝F，t　×　xioutiet／x，toatiet

By　the　same　way，　methane　conversion　（C．．th．，　）　can　be　calculated　as，

C蜘。一（Fm。，h。。。u’・t－F　、h　戚let）／F蜘。畦1－Fm。，h。。。幽Fm。・h＿’niet

＝1－FAr　×　XmethaneOntlet／XArO”tlet　×　F．．th．．，inlet

　　　　　The　gaseous　p　roducts　given　off　from　the　cat　aly　st　and　heat　effect，　weight　change

of　the　cataly　st　in　the　carburization　p　rocess　were　monitored　with　a　DTA－TG－M　S

combined　instrurnent（Mas　Science　DTA－2020S＋Thermolab）。　The　e脚ment　wお

performed　with　about　40　mg　coked　cat　aly　st　p　aced　in　a　alumina　contain　er　un　der　fiowing

150　ml／min　H　e　and　15　ml／Min　methan　e　with　ramping　the　temperature　of　5　K／min．

　　　TPSR　（T　emp　erat　ure　P　rogrammed　Surface　React　i　on）　exp　eri　ments　was　p　erformed

with　about　200mg　cat　aly　st　p　laced　in　a　U－shap　e　quartz　reactor．　Cat　aly　st　zone　was

heat　ed　in　a　He　stream　to　873　K　and　held　at　873　K　for　30　min　then　cooled　to　room

temp　erature．　At　room　temp　erat　ure　the　stream　was　switched　to　methane　feed．

Temp　erature－programmed　methane　react　ion　was　started　from　room　temp　erature　to　973

K　at　10　K／miri　and　then　hel　d　at　973　K　for　30　rnin．　The　p　roduct　s　such　as　H2　（m／e＝　2），

CO（m／e＝＝28），　C　O　2（m／e＝44），　C2H4　（rn／e＝26），　C2H6　（m／e＝30），　C6H6　（rn／e＝78）　was

monitored　with　TE－150　multip　le　channel　mass　spectrometer　and　the　spect　ra　was

recorded　at　the　same　ti　me．　Reaction　pressure　is　34　kPa．
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3　Results　and　discussion

3．1　Reduction　performance　of　Re／HZSM－5　in　methane　and　in　hydrogen　flow

　　　Cataly　t　ic　features　of　Re207／HZ　SM－5－based　cataly　st　for　activat　ing　methane　were

investigat　ed　by　T　G－DTA－M　S　meas　urement．　Vthen　Re207／HZ　SM－5－bas　ed　cataly　st　was

87
5
r
r
」
3
1
．

・ピ

@　　　　．：1，］1，・・一リ

　　　　ユ　　ドマニリロ

・’ O弦蘇．
　　　　　　　　　，ζ∫’ギ∫



heated　in　methane　flow　by　p　ro　gra㎜med　temperature丘om　673　to　973　K，　reduction

process　of　Re207／HZ　SM－5　by　methane　first　take　p　lace　at　about　873　K，　and　then

benzene，　naphthalene　and　ethene　were　produced　on　the　reduced　caね1yst　shortly　as

shown　in　figure　1．　ln　the　reduction　p　rocess　of　Re207！HZ　SM－5　by　methane，　CO，　H20，

H2　and　a　little　C2H6　were　given　off，　and　a　strong　endothermic　effect　comp　any　ing　the

reduction　were　identified　by　DTA　as　shown　in　fig　2．　This　is　quite　different　from　the

reduction　of　Re207／HZ　SM－5　by　H2．　lt　is　shown　for’the　reduct　ion　p　rocess　of

Re207／HZ　SM－5　by　H2　in　fig　3　that　the　reduction　starts　at　ahout　573　K　and　it　is　a

exDthemi　c　p　rocess．　T　o　exp　Iain　the　different　reduct　ion　p　erforman　ce　of　Re207　in

methane　and　in　hy　drogen，　we　p　rop　ose　here　that　a　new　phase　of　rhenium　carbide　is

formed　by　reduction　of　of　RchO7　in　methane　as　shown　in　scheme　1．　The　carbide　is

stable　in　the　atmosphere　of　methane　and　is　active　to　cataly　ze　the　aromatization　reaction

of　methane．　The　temp　erature　for　formation　of　rhenium　carbide　is　about　100　K　lower

than　that　for　moly　bdenum　carbide　formation　at　the　same　conditions，　therefore，　low－

temperature　activity　of　Re／HZSM－5　is　better　than　that　of　Mo／HZSM－5－based　catalyst．

3．2．　Catalyti　c　performance　of　RelHZS　M－5　for　dehydro－di　merizati　on　and　dehydro－

aromatization　of　methane

　　　The　act　ivity　and　the　selectivity　of　Re／HZ　SM－5－based　cataly　st　with　different　Re－

loading　for　dehy　dre－dimerizat　ion　and　dehy　dro－aromat　izat　ion　of　methane　at　873　K　is

listed　in　table　1　and　comp　ared　with　that　of　M　o／HZ　SM－5－based　cataly　st．　100／o

Re／HZ　SM－5　shows　high　benzene　selectivity　and　low　coke　selectivity．　lri　addition，　to

co　mp　are　with　Mo／HZ　SM　一5－bas　ed　cat　aly　st，　Re／HZ　SM　一5－bas　ed　cat　aly　st　exhibits

relatively　much　higher　activity　and　selectivity　for　dehy　dro－dimerization　．of　methane　to

form　C2　hy　drocarbons　and　lower　activity　and　selectivity　for　nap　hthalene　p　roduction．

The　bimetal　catalyst　of　Mo－Re／HZSM－5　shows　a　higher　C2　selectivity　than

Mo／HZSM－5　and　a　higher　naphthalene　selectivity　than　ReLHZSM－5．

　　　　Dep　endence　of　act　ivity　and　select　ivity　of　ReZHZ　SM－5　cat　aly　st　on　react　ion
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temp　erature　and　specific　velocity　is　given　in　tahle　2　and　fig　4，　resp　ectively．　M　ethane

oonverslon　and　benzene　selectivity　increase　sharp　ly　with　increasing　temp　era加re　ftom

873　to　1023　K　as　shown　in　fig　4．　Nap　hthalene　selectivity　gn　the　maximum　value　at

973　Kand　then　decrease　with　further　hlcrease　of　tempera加re．　Low　reaction

temperaれrre　lead　to　a　high　C2　selectivity．

　　　Benzene　formation　rate　can　be　great　1y　enhanced　by　increasing　specific　velocity　of

methane．　However，　cataly　st　deactivates　quickly　at　very　high　sp　ecific　velocity　of

methane．　So　benzene　format　ion　rate　can　not　be　enhanced　further　by　increas　ing　SV　from

5000　to　10000　ml／h／g－cat　as　show　in　fig　4．

3．3　Effect　of　H2　pretreatment　on　catalytic　performance　ef　ReNZSM－5

　　　Fig　5　shows　the　effect　of　p　retreating　cataly　st　by　hy　drogon　and　methane　on　the

activity　of　Re／HZ　SM－5　for　benzene　format　ion．　lt　is　shown　in　fig　5　that　p　ret　reatment　of

Re／HZ　SM－5　by　hy　drogen　an　d　methane　lead　t　o　decreas　e　of　the　start　ing　t　emp　erat　ure　for

benzene　format　ion．　The　p　ret　reat　ed　Re／HZ　SM－5　by　hy　drogen　at　a　lower　ternp　erat　ure　of

5730r　623　K　shows　the　best　low－tempe耐ure繊i酒ty　for　benzene　formation　and　the

startingtemperature　for　prbducing　benzene　is　about　773　K．　The　time　on　stream　results

in　fig　6　also　indicate　that　the　stability　and　activity　of　Re／HZ　SM－5　cataly　st　for　benzene

formation　is　enhanced　by　H2　p　retreaiment．　This　is　quite　different　from　the　case　of

Mo／HZSM－5　type　catalyst．

　　　　　　　The　p　roducts　for　the　temp　erature－p　rogrammed　reaction　of　methane　on

Mo／HZ　SM－5　are　identified　by　a　mass　sp　ectrometer　and　shown　as　in　fig　7．　The　mitial

temp　erat　ure　for　benzene　format　ion　is　as　high　as　about　950K．　Pret　reat　ment　by　hy　drogon

reduct　ion　and　carburization　also　lead　to　decrease　of　the　init　ial　temp　erat　ure　for　benzene

formation　as　shown　in　fig　8．
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3．4　Copling　effect　of　Co　in　R“Co／HZSM－5　catalyst

　　　The　activity　of　Re侃ZSM－5　and　Re－Co／HZSM－5　for　benzene　formation　with　time
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on　stream　was　comp　ared　in　figs　9－11．　lt　is　indicat　ed　in　figs　9－11　that　the　activity　of

Re／HLZ　SIvl－5　is　evi　dent　ly　enlian　ced　by　addit　ion　of　O．50／o　Co　to　100／oRe／HZ　SM－5　at

various　reaction　conditions．　However，　the　Re－Co／HZ　SM－5　cataly　st　is　not　very　stable

with　time　on　stream　in　the　reaction　conditions．

4　Conclusion

　　　Re207／HZ　SM－5　is　endothermically　convert　ed　t　o　its　reduced　stat　e　heat　and　it

accomp　any　with　p　roduction　of　COx　and　loss　of　cataly　st　weight　in　methane　flow　before

product　ion　of　hy　drocarbon　p　roducts．　The　t　emp　erature　for　forrnation　of　rhenium　act　ive

phase　is　about　100　K　lower　than　that　for　moly　bdenum　type　cataly　st．　As　a　result，

Re／HZ　SM－5－based　cat　aly　st　exhibits　a　better　low－temp　erature　activity　and　a　higher

selectivity　to　C2　hy　drocarbons．　ln　addition，　activity　and　stability　of　Re／HZ　SI）v｛一5－

based　cataly　st　is　also　dep　endent　on　p　retreatment　conditions　or　addition　of　modifier　such

as　Co　to　it．　The　p　retreatment　of　Re／HZ　SM－5　by　hy　drogent　lead　to　a　better　cataly　tic

activity　and　stability．　Addition　of　Co　to　Re／HZ　SM－5　is　effective　to　enhance　the

actlvlty．
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　　　　　　　　Temperature／K

TG－DTA－MS　spectra　of　Re／HZSM－5　in　hydrogen　flow
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Table　1．　Activity　and selectivity　of Re1HZSM－5 catalysts for

dehydro－dimerization　and　dehydro。aromatization ト
・㌔

ﾙ

Catalyst　　　　　　　　Conve：rsion Selectivity fbr　product　fbrmation
（％）

1

，　，

1

（％） Ben．　To1． Naph． C2 CO Coke

3％Re／HZSM－5　　　　2．7 29．9　3．1 4．5 6．5 1．4 51

10％Re1HZSM－5　　　2．1 45．3　5．4 3．9 13．5 1．8 30

20％Re1HZSM－5　　　1．9 38．3　4．5 2．5 12．1 1．9 40
1

3％Mo／HZSM・5　　　　2．6 35．4　3．3 17．6 6．1 0．1 37

15％Mo／HZSM－5　　　2．8 32．2　2．8 16．7 5．4 5．4 38

3％Re3％Mo／HZSM－5　2．5 36．3　4。1 11．6 7．9 0．7 39

Temperature＝873　K，　SV＝1440 ml！h／9－cat，　P＝ 3atm
・

Ta『ble　2．Dependence　of　activity　and　selectivity onre temperature

for　dehydro－dimerization　and dehydro－aromatization
●　　，

Tempe■ature　　　Conversion Selectivity fb：r　product　formation （％）
1・

（K）　　　　　　　（％） Ben．　To1． Naph． C2 CO Coke

873　　　　　　　　2．1 45．3　5．4 3．9 13．5 1．8 30 ，

923　　　　　　　　4．2 43．0　　3．8 8．3 5．9 0．9 42 ，

973　　　　　　　　6．8 47．6　3．3 11．4 3．9 0．2 33
r

1023　　　　　　　9．6 50．6　3．3 3．0 3．8 0 39
♂

1．

SVニ1440　mVh／g－cat，　P・＝3atm， catalyst：10％ Re／HZSM－5
o，

f－

s・ ﾑ、・

@　、1・’
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According　to　the　obtained　results，　the　process　for　formatio　of

rhenium　carbide　active　phase　and　for　production　of　C2　and
　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　のaromatics　by　dehydro－dimerizatio　and　dehydro－aromatlzatlon

of　methaneis　schemed　as　follows，

ド

」
重
臣
砦

CH4

Re207／HZSM－5

co
H2
H20
C2H6

ReC／HZSM－5

1　，dchyd　ro－di　meri　zati　on
cH4　hH2　c｛H4
　　　　　ReC

H－transfer　reaction

．ptl；．：：’fr．一d一一．．　C　2H6

　ReC

2，　dehydro－aromatization 十 CioHs
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　473　573　673　773　873　973

　　　　　　　　　　　Temperaturei［K

1，　fresh　catalyst　2，　pre－carburized　at　953K　3，　pre－re　duced　at　9　73K

TPR　profiles　for　benzene　formation　on　30／o　Mo／HZSM－5　pretreated　at

diliferent　conditions
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Effect　of　Co　addition　to　Re／HZSM－5　on
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Chapter　9
Methane　conversion　with　CO　into　aromatics　on　Re／HZSM－5　catalyst

Abstract

Catal　yti　c　com研sion　of　methane　With　CO　to　aromaties　on　Re1HZSM－5－based

catalyst　are　studied　Both　methane　and　CO　are　consumed　and　contri　beted　to

the　production　of　aromatics．　The　reaction　of　methane　with　CO　is　expressed　as：

3CH4＋3co　一　Q　＋　3H20

4（H4＋4CO　一一一一i）＞O　＋　2CO2　＋　5H2

Q一一・一（（）（）
〉

C（coke）　＋　CO2　．

coke

2CO

Benzene　production，　therefore，　is　greatly　enhanced　by　adding　CO　to　methane

feed．　il　ad｛血tion，　the　conversion　of　CO　With　methane　also且ead　to　fbrmation　of

some　CO2，　which　reacts　with　coke　to　eliminate　it．　Therefore，　the　coke　deposition

on　the　catalyst　is　decreased　and　catalyst　stability　is　enhanced　by　CO　additien．
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1　lntroduction

　　　　The　catalytic　features　of　Re／HZSM－5－based　catalysts　for　dehydrogenation　and

aromatization　of　methane　was　presented　previously．　Compared　with　Mo／HZSM－5，

Re／HZSM－5　shows　much　better　lower－temperature　activity　for　production　of　aromatics．

In　the　present　research，　methane　conversion　with　CO　to　aromatics　on　Re／HZSM－5　is

studied　and　a　mechanism　for　the　reaction　is　suggested．

2　Experimental

　　　Re／HLZSM－5　was　prepared　by　impregnating　NH4ZSM－5　（SiO2／AI203＝40，　Toso

Co．）　with　NHUReO44H20　（Kanto　Chem　Co．）　aqueous　solution，　followed　by　drying　at

383　K　for　4　h　and　calcination　at　773　K　for　4　h．　The　sample　was　fmally　pressed，

crushed　and　sorted　into　20－40　mesh．

　　　　　Catalytic　tests　were　carried　out　with　about　500　mg　catalyst　placed　in　a　fixed　bed

continuous－fipw　straight　type　quartz　reactor．　The　raw　materials　of　methane　and　CO　iS

over　99．990／o　pure．　All　the　flow　rate　of　gas　was　controlled　with　mass　fiow　controllers

（ASAHI　Keiki　Co．　LTD）．　The　products　were　withdraw　periodically　from　the　outlet　of

reactor　and　analyzed　by　two　on－1ine　gas　chromatography　of　Shirnadazu　oc－9A　and　（］K）一

8A，qゆP・d・With・a・fl・m・i・nizati・n　d・tect・・（FID）and・a・th・rmal。・nduc鵬detecto「

（TCD）respectively．　Hydrocarbon　products　were　separated　by　a　4　mm　×1m　Porapak　P

column　and　Ar　，　CH4，　CO，　CO2，　H2　were　analyzed　by　a4　mrnxlm　active　carbon　colurm．

10．70／o　Ar　in　methane　was　emp　loy　ed　as　internal　standard．　Formation　rate　of　p　roducts，

methane　conversion　were　calculated　by　an　int　ernal　standard　analy　zing　method　described

as　previously．　Below，　the　method　is　presented　briefly．　Even　if　the　total　flow　rate

changed　at　outlet　from　inlet　of　reactor　（FT．ulintiCt　，FT．taiO”tiet　）．　Flow　rate　of　non－reactive

intemal　standard　gas　（F，t）　doses　not　change．

　　　F，t　＝FT．t．iO”tiet　XX，tOutiet　＝　FT．t．iinict　xx，tinlet

Here，　X，t　refer　to　concentration　ofthe　intemal　standard　gas．

Forrnation　rate　of　i　product　（FD　can　be　calculated　as，

F董一F，．，。、・utlct×Xieutlet－F，，×Xi戚1．t灰，，．面d

魅
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By　the　same　way，　methane　conversion　（C．．th．，　）　can　be　cal　culated　as，

C蜘。一（Fm．，h＿u…一Fm。，h＿副d）／F蜘。h’Ct・・　1・F蜘．噛Fm．伽。．沁1d

＝1－FAr　×　XmethatteOutlet／XArOUtlet　×　Fm．th．neinlet

CO　conversion　（Cco）　is　calculated　as，　Cco　＝（　Fcoiniet　一　FcoOatict）／　Fcoinict

HC　se1戯ivity　is面。司ated　as　SH♂formation耐e　of　HC／（consurnption　rate　of　methane

＋　consumption　rate　of　CO）

3　Results　and　discussion

3．1　Featu　res　of　methane　reaction　with　CO　on　Re／HZSM－5

　　　　The　consump　t　ion　rat　es　of　CO　and　methane　as　well　as　the　format　ion　rates　of

products　with　time　on　stream　for　methane　conversion　with　CO　on　Re／HZ　SM－5　are

shown　in　figs　1　and　2．　The　main　p　roduct　s　for　reaction　of　methane　and　CO　on　the

cat・aly・t　are　b・nzene，　n・phthal・n・，　C、　and　a　littl・CO・・Th・。・nsumpti・n耐㏄・f

methane　and　CO　initially　is　higher　and　then　kep　t　to　be　constant　with　time　on　stream

aft　er　about　2　h．　Format　ion　rat　e　of　benzene　is　lower　and　CO2　format　ion耐e　is　higher　in

the　initial　stago　and　then　kept　to　be　stable．　in　addition，　formation　rate．　of　CO2　is　much

le＄s　than　a　half　of　CO　consumption　rate，　Active　carbon　species　and　CO2　is　p　roduced

from　B　udart　reaction　of　CO．　We　sup　p　osed　that　the　active　carbon　sp　ecies　is　contributed

to　the　p　roduction　of　higher　hy　drocarbons　by　reaction　with　hy　drogpm　or　with　methane

which　p　rovide　hy　drogen　source．　On　the　other　hand，　coke　deposition　on　the　cataly　st，

vvhich　lead　to　decrease　of　cat　aly　st　activity，　is　inhibited　by　the　react　ion　of　coke　with　CO2

to　p　roduce　CO　（coke　＋　CO2　一一　2CO）．　Therefore，　the　amount　of　p　roduced　CO2　is　much

less　th　an　what　expected　from　B　udart　reaction　（　about　a　half　of　consumed　CO）．

3．2　Effect　of　COImeth　ane　rati　o

　　　　Dep　endence　of　benzene　format　ion　rat　e　on　the　concentrat　ion　of　CO　in　methane　feed

is　shown　in　fig　3．　Rat　e　of　benz　ene　forrnat　ion　is　great　ly　enhanced　and　stabilized　by

increasing　the　concentration　of　CO　in　methane　from　O　to　22．20／o．　Further　incre2ise　of
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CO　from　22．2　to　47．70／o　lead　to　decrease　of　cataly　st　stability．　Conversion　of　methane

and　selectivity　of　total　hy　drocarbon　p　roducts　increase　by　addition　of　CO　to　methane

feed　as　shown　in　fig　4．　Therefore，　both　methane　and　CO　are　contribut　ed　to　p　roduction

of　higher　hy　drocarbons．　lt　also　can　be　seen　from　fig　4　that　CO　convers　ion　increases

with　increas　e　of　CO　concent　rat　ion　in　methane　feed．　Select　ivity　of　total　hy　drocarbon

produCts　is　maximized　at　a　CO　concent耐ion　of　about　8．6％and　benzene　format　ion　rat　e

is　maximized　at　a　CO　concentration　of　about　22．20／o．

3．3　Effect　ofvarying　methane　SV　and　reaction　temperature

　　Dep　endence　of　conversion，　selectivity　and　benzene　formation　rate　on　react　ion

temp　erature　is　shown　in　fig　5．　Both　methane　conversion　and　selectivity　of　total　higher

hy　dro　carbon　p　roduct　s　are　rernarkahly　enh　anced　by　increas　ing　react　ion　t　emp　erat　ure　from

973　K　to　1073　K，　although　CO　conversion　decreases　a　little　at　higher　temp　erature．

Benzene　format　ion耐e　is　almost　doubled　by　inereasing　t　emp　erat　ure　from　973　K　to　1023

K．　This　indicates　that　higher　reaction　temp　erat　ure　is　more　favorable　for　the　catalytic

conversion　of　methane　with　CO　to　higher　hy　drocarbon　p　roducts　on　Re／HZ　SM－5．　Fig

6　shows　that　Re／HZ　SM－5　cataly　st　for　the　reaction　at　high　temp　erature　is　quite　stable

with　t　ime　on　stream．　Rate　of　benzene　format　ion　increases　and　nap　hthalene　is　decreases　a

little　with　time　on　stream　Methane　oonversion　almost　k㏄p　Oonsta耐within　500　min．

Effect　of　methane　specific　velocity　on　conversion　of　methane　and　CO，　format　ion　rat　e　of

benzene　and　selectivity　to　total　hy　drocarbon　p　roducts　is　shown　in　fig　7．　Benzene

fo　rmat　ion耐e　in（rreases　by　a　order　of　m～ignitude　by　increasing　methane　SV　from　1440　to

10000rn1／h／g－cat．　However，　methane　oonversion　changβlittle　with　increasing　methane

SV．　T　herefore，　Re／HZ　SM－5　cataly　st　exhibits　very　high　activity　for　methane

aromat　ization　with　CO　at　high　SV．　Fig　8　indicates　that　a　lower　CO　concentration　of

about　8．60／o　in　methane　feed　lead　t　o　a　higher　act　ivity　and　stability　for　benzene　forrnat　ion

at　high　methane　specific　velocity．
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3．4　The　mechanism　suggested　for　the　catalytic　conversion　of　methane　with　C　O

　　　　Accordmg　the　analysis　of　products　and　consumption　of　methane　and　CO，　the

conversion　of　methane　with　CO　on　ReiHZSM－5　proceed　as　follows，

3CH4＋3CO　一一一．一　0　＋　3H20

4CH4＋4CO　一＞Q　＋　2CO2　＋　5H2

Q　一一’一〇〉　co　一〉　coke

C（coke）　＋　CO2　一）〉　2CO

Contribution　of　CO　on　the　p　roduction　of　aromatics　and　p　roduction　of　CO2　are　described

by　following　mechanism，

　　Mechanism：

clt レ　CHx レ（hK レ◎ レ◎つ

監
■
菖
蟹
匡

co＋rp　＝〉　clk

　　　　　　　　　　　　－o，　CQ2

C　（coke）　＋　CQ レ　2CO

The　role　of　CO　addition　for　stabilizing　cataly　st　is　due　to　the　reaction　of　p　roduced　CO2

with　coke　to　produce　CO．　Therefore，　coke　deposition　on　the　catalyst　is　decreased．
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4　Conclusion

　　　　Act　ivity　and　st　ability　of　Re／HZ　SM－5　cat　aly　st　for　methane　aromat　iz　at　ion　is

greatly　enhanced　by　CO　addition　to　methane　feed．　CO　is　contributed　to　the　formation

。f　benzene　by血te・acti・繭th　me伽e・n　Re／HZSM－5翻yst　I副diti・n・alittle

CO2　is　also　produced　as　a　side　product，　which　may　play　a　role　to　enhance　the　catalytic

stahility．
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Fig　1．　Consurnption　of　CH4　and　CO　and　formation　rate　of　CO2　with　time　on　stream

for　methane　conversion　with　C　O　on　100／o　Re／HZ　S　M－5

Fig　2．　Conversion　and　product　formation　rate　with　time　on　stream　for　methane

conversion　with　22．20／o　CO　on　100／oRe／HZ　S　M－5　at　1023　K　and　3　atm

Fig　3．　Benzene　forrnation　rate　with　time　on　stream　for　methane　conversion　with

different　CO　concentration　on　100／oRe／HZSM－5

Fig　4．　Conversion　of　CH4　and　CO，　Selectivity　of　higher　hydrocarbon　products　and

formation　rate　ofbenzene　as　a　function　of　CO　concentration　in　methane　feed

Fig　5．　Conversion　of　CH4　and　CO，　Selectivity　of　higher　hydrocarbon　products　and

formation　rate　of　benzene　as　a　function　of　reaction　temp　erature

Fig　6．　Conversion　and　formation　rate　of　products　with　time　on　stream　for　methane

conversion　with　22．20／o　CO

Fig　7．　Conversion　of　CH4　and　CO，　Selectivity　of　higher　hydrocarbon　products　and

formation　rate　ofbenzene　as　a　function　of　methane　specific　velocity

Fig8．　Benzene・f・・mati・n耐e輌thtim・・n・stream　for　methane。・nversi・n　With　diffe「ent

CO　concentration　at　a　higher　methane　SV
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Fig　1．　Consumption　rate　of　CH，　and　CO　and　formation　rate　of　CO2　with

time　on　stream　for　methane　conversion　with　CO　on　10‘）6　Re／HZSM－5
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Fig　2．　Convefsion　and　formation　rate　p．f　prodyclt＄　yyitLh一！一iq）e－on　stream

foi　methane　conversion　with　22．2％　CO　on　1096　Re／HZSM－5
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Fig　3．　Benzene　formation　rate　with　time　一〇n　stree－i11．　一forh．一m．Ae．t．h．a．　ne

coV獅魔?ｒｓｉｏｎ　with　difterent　CO　concentration　onlO96Re／HZSM－5
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Fig　5．　Conversion　of　CH4　and　CO，　selectivity　of　higher　hydrocarbon　products

and　formation　rate　of　benzene　as　a　function　ot　reaction　temperatures
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Fig　8．　Benzene　formation　rate　with　time　on　stream　for　methane　conversion　with
CO　of　deifferent　concentration　on　1096Re／HZSM－5　at　a　higher　SV　of　methane
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Chapter　10
Me伽ne　aromatiZation　with　CO2　on　Re　ltHZSM－5　c漁lyst

Abstract

Effect　of　CO2　addit　ion　on　aromatization　of　methane　on　Re／HZ　SM－5－bas　ed　cat　aly　st　are

studied．　Selectivity　of　hy　drocarbon　p　roduct　and　format　ion　rat　es　of　benzene　and

nap　hthalene　are　great　ly　enhanced　by　addit　ion　of　about　1－30／o　C　O2　t　o　methane　feed．

Selectivity　for　coke　dep　osition　is　about　O　at　the　p　resence　of　1－30／o　CO2　in　methane　feed．

Over　900／o　of　CO2　is　consumed　and　transformed　to　CO　by　reaction　with　coke．　The

reaction　of　methane　with　C　O2　is　expressed　as：

㎝4

　　　　　　　CO　2　＋　Coke　一〉　2CO

　　　　　　　　CO＋　H2　一〉　CH．＋　H20

1n　addition，　amount　of　p　roduced　CO　is　less　than　2　times　of　consumed　CO2，　with

methane　also　lead　to　formation　of　some　CO2．　Therefore，　a　p　art　of　p　roduced　CO　is

secondarily　react　with　hy　drogon　to　p　roduce　hy　drocarbon　p　roducts，　which　is　responsible

for　the　increase　of　hydrocarbon　selectivity．
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1　lntroduction

　　　　　The　catalytic　features　of　Re／HZS　M－5－based　catalysts　for　dehydrogenati　on　and

aromatization　of　methane　and　effect　of　C　O　addition　to　methane　feed　on　the　catal　ytic

reaction　were　discussed　prev　iously．　Compared　with　Mo／HZS　M－5　［1－20］，　Re！HZS　M－5

shows　much　better　lower－temperature　activity　for　prod　uction　of　aromatics　and　d　i　fferent

selectivity　to　C，　and　naphthalene　［21］．　I　n　the　cas　e　of　C　O　ad　dition　to　methane　feed，　both

methane　and　CO　are　consumed　and　contributed　to　prod　uction　of　aromatics　on　Re／HZS　M－

5　［22］．　In　the　present　res　earch，　effect　of　C　O，　addition　to　methane　on　methane

aromatization　on　Re／HZSM－5　is　studied．

2　Experimental

　　　　　　Re／HZ　SM－5　was　p　rep　ared　by　imp　regn　at　ing　NH4Z　SM－5　（SiO2！AI203＝40，　T　oso

Co．）　with　NH4ReO44H20　（Kanto　Chem　Co．）　aqueous　solution，　followed　by　drying　at

383　K　for　4　h　and　calcination　at　773　K　for　4　h．　The　samp　le　was　fmally　p　ressed，

crushed　and　sorted　into　20－40　mesh．

　　　　　Cat　aly　t　ic　t　ests　were　carri　ed　o　ut　with　ahout　500　mg　cat　aly　st　p　laced　in　a　fixed　bed

continuous－fiow　straight　ty　p　e　quartz　reactor．　The　raw　materials　of　methane　and　CO2　is

over　99．990／o　p　ure．　All　the　fiow　rat　e　of　gns　was　controlled　with　mass　flow　cont　rollers

（ASAHI　Keiki　Co．　LTD）．　The　p　roducts　were　withdraw　p　eriodically　from　the　outlet　of

reactor　and　analy　zed　by　t　wo　on－1ine　gas　chromatograp　hy　of　Shimadazu　cc－9A　and　oc－

8A　eqゆped　With　a　flame　ionization　detector（FID）and　a　thermal　conductiVity　detector

（TCD）respectively．　Hydrocarbon　products　were　separated　by　a4　mm　×lm　Porapak　P

colurrm　and　Ar　，　CH4，　CO，　CO2，　H2　were　analy　zed　by　a　4　mmx　l　m　active　carbon　column．

IO．70／o　Ar　in　methane　was　emp　loy　ed　as　int　ernal　standard．　Format　ion　rat　e　of　p　roducts，

methane　conversion　were　calculated　by　an　internal　standard　analy　zing　method　described

as　p　reviously．　Below，　the　method　is　p　resented　briefly．　Even　if　the　total　flow　rate

charigpd　at　out　let　from　inlet　of　reactor　（FT．laiintiC‘　，FT．otO“tiet）．　Flow　rate　of　non－reactive

intemal　standard　gas　（Fst）　doses　not　change．

　　F，監＝FT伽10瓜1et×X，こ。utlet＝FT。t．li「tlじt×X、竃inld

t
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Here，　Xst　refer　to　concentration　of　the　intemal　standard　gas．

Formati　on　rate　of　i　product　（FD　can　be　cal　cul　ated　as，

Fi＝FT．t．10”tlet　X　XiOutlet　＝＝Fst　×　xioutlct　ltxstoutlet

By　the　same　way，　methane　conversion　（C．．th．．　）　can　be　cal　culated　as，

Cmcthane　＝　（Fmethaneiniet　一FmethaneOUtlet）／　Fmethaneinlet　＝　1一　FmethaneOutlet／　Fmethafieialet

＝1－FAr　×　XrnethaneOut；et／XArOutlct　×　Fm．th．n．inlet

CO2　conversion　（Cco2）　is　cal　cul　ated　as，　Cco2　＝（　Fco2iniet　一　Fco20utiet）／　Fco2iniet

HC　selectivity　is　cai　culated　as　SHc＝　formation　rate　of　HC／consumption　rate　of　methane

Coke　selectivity　＝　（　consumed　CH4　＋　consumed　CO2　一produced　HC　一　p　roduced

CO）／consumed　methane．

3　Results　and　discussion

3．1　Achievement　of　high　and　stable　benzene　formation　by　addition　of　CO2

　　　Benzene　formation　rate　with　time　on　stream　is　very　stable　and　a　stable　formation

rat　e　of　benzene　is　about　130　nmolels／gcat　for　methane　aromat　izat　ion　on　Re／HLZ　SM－5

cat　aly　st　at　a　methane　SV　of　1440　ml！h／gcat．　A　higher　rate　of　benzene　format　ion　can　be

obtained　by　increasing　methane　specific　velocity　．　However，　benzene　formation　rate　with

time　on　stream　is　not　stable　as　shown　in　fig　1．　Benzene　formation　rate　at　the　initial

stage　is　about　500　nmole／s／g－cat　at　a　met　h　ane　SV　of　5000　mlth／g　cat　an　d　decreas　es　very

quickly　with　time　on　stream．　To　solve　this　p　roblem，　some　CO2　with　different

concent　rat　ion　is　added　to　methane　feed．　A　high　and　stable　rate　of　benzene　formation

with　time　on　stream　has　been　attained　for　methane　aromatization　with　adding　a　little

CO2　t　o　methane　feed　on　Re／HZ　SM－5　at　a　high　methane　SV　of　5000　ml／h／gcat　as　shown

in　fig　2．　1－30／o　CO2　addition　in　methane　feed　is　imp　ortant　for　obtaining　a　high　a　stable

format　ion　rate　of　benzene．　A　higher　CO200ncent耐ion　of　8．6％in　methane　feed　lead　to　a

suppression　of　benzene　formation　rate．　ln　addition，　benzene　format　ion　rate　can　be

increased　further　by　increasing　methane　SV　and　reaction　temp　erature，　About　1600

nmole／s／gcal　of　benzene　formation　rat　e　is　achieved　for　methane　aromat　izat　ion　with
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about　2．IO／o　CO2　on　100／oRe／HZSM－5　at　1073K　and　methane　SV　of　10000　ml／h／g－cat．

3．2．　Conversion　of　CH4　and　CO2　and　analysis　of　products

The　results　for　conversion　of　CH4　and　CO2　and　analysis　ofproducts　by　employ　ing

CH・輌th　2％CO・お鋭d　a・・h・㎝㎞鯵4㎝d　5．　It　is　sh・㎜㎞丘9・4・that。v，，90％

CO2　is　converted　to　CO．　Formation　rate　ofCO　is　more　than　consumption　rate　of　CO2，

but　less　than　2　times　of　CO2　cosumption　rate．　This　indicate　that　some　methane　is　also

。ont「ibuted　t・the　f・磁i・n・f　CO・It　is　sh・wn　in　figsthat　maj・r　hydr・ca・b。n　pr。duα

is　benzene　and　f・職1・蝋e・f　naphalene，　C2　and・T＋X　are・much・1ess・than・f。噸1。n

rate　of　benzene．　ln　addition，　it　is　also　clear　in　fig　5　that　methane　conversion　is　high　and

benz　ene　formation　rat　e　is　lower　in　mitial　stage　and　then　become　stable　with　time　on

stream．　Therefore，　there　is　a　introduce　period　for　methane　conversion　with　CO2．

3．3　The　role　ofCO2　in　enhancing　HC　selectivity　and　inhibiting　coke　deposition

Product　selectivity　is　analyzed　and　results　are　shown　in　fig6．　lt　is　clear　in　fig　6　that

the　selectivity　ofhydrocarbon　pToducts　js　evidently　enhanced　by　add」tion　of　1－20／o　CO2　．

It　is　quit・ゆ・rt・nt　that　selectivity　fo・。・k・d・p・・iti・n・n　th・剛y，t　i，　ah。ut　O　wh㎝

1－30／o　CO2　is　added　to　methane　fee（3　as　shown　in　fig　6．　in　addition，　benzene　formation

rate　is　maximized　at　1・一20／o　CO2　in　methane　feed．　Higher　concentration　of　CO2　，　for

mstance，　8．60／o　CO2　in　methane　feed　Iead　to　a　sharp　sup　p　ression　of　benzene　formation．

As　s・・n　as。・k・is　d・p・・ited・n　th・catalyst　fr・m　m就h㎝・ar・職izati。n，　i圃1，eaCt

with　CO2　to　produce　CO　through　reaction：　CO2　＋　C（coke）　＝＝　2CO．　This　reaction　is

responsible　for　inhibition　of　coke　deposition　on　the　catalyst　and　that　the　coke　selectivity

is　about　O．　On　the　other　hand，　the　produced　CO　may　further　react　with　hy　drogen　and　is

partially　contributed　to　the　production　ofbenzene．　Therefor，　selectivity　of　hydrocarbon

products　is　high　when　1－20／o　CO2　is　added　to　methane　feed

’

3．4　Effect　of　reaction　temperature　and　methane　SV
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The　results　in　fig　7　indicate　that　CH4　conversion　and　benzene　format　ion　rat　e　increase

remarkably　with　increase　of　reaction　temperature．　HC　selectivity　and　CO2　conversion

are　maximized　at　a　react　ion　t　emp　erat　ure　of　1023　K．　At　a　s　elect　ed　reaction　t　emp　erat　ure　of

1023　K，　methane　conversion　with　2．10／o　CO2　is　conducted　by　varying　methane　specific

velocity．　Conversion　of　CH4　and　CO2，　selectivity　of　hydrocarbon　products　and

f・臓i・n耐・・fb・nzen・おa㎞αi・n・fm就h㎝・SV　is　sh・wn　in　fig　8．　lt・is・indi（綱in

fig　8　that　CH4　conversion　chango　little　with　increase　ofmethane　SV．　However，　benzene

f・職i・n・at・in・・eases　by・ne・・de・・f　magnit　ud・by　in　creas　ing　met　hane　S＞丘。m　l440

to　10000　mlZh／g－cat．　HC　selectivity　and　CO2　conversion　also　increase　a　little　by

mcreasing　methane　SV．　So，　Re／HZSM－5　catalyst　is　quite　active　for　methane

conversion　with　C　O2．

3．5　The　mechanism　for　the　reaction　of　methane　with　CO2

　　　Based　on　the　above　results，　the　role　ofCO2　in　enhancing　selectivity　of　hydrocarbon

products　and　inhibiting　coke　deposition　is　described　as　below

　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　，

CH4　’?’　CHx

H2

＞c29一
　　　　　　　　　　　H2

〉　Coke

COっ　＋　　Coke 〉　2CO

　　　　　　　　　CO＋　H2　一〉一　CH．＋　H20

The　reaction　of　CO2　with　coke　to　produce　CO　and　the　secondary　reaction　of　produced

CO　with　hy　drogen　to　produce　CHx　species　are　responsible　for　the　low　coke　selectivity

and　high　hydrocarbon　selectivity．

4　Conclusion

　　　High　and　stable　rate　of　benzene　formation　is　achieved　at　a　high　SV　of　methane　on

Re／HZSM－5　type　catalyst　by　addition　of　about　1－30／o　CO2　to　methane　feed．　A　little
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amount　of　CO2　in　methane　feed　is　very　effective　to　sup　p　ress　coke　dep　osition　on　the

cataly　st　and　selectivity　ofhy　drocarbon　p　roduct　s　is　greatly　imp　roved　by　addition　of　CO2

to　methane　feed．　The　reaction　of　CO2　with　coke　to　p　roduce　CO　and　the　secondary

react　ion　of　p　roduced　CO　with　hy　drogen　to　p　roduce　CHx　sp　ecies　are　responsible　for　the

low　coke　selecti　vity　and　high　hydrocarbon　selectivity．
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Caption　of　figures：

Fig　1．　Benzene　formation　rat　e　with　time　on　stream　for　methane　aromatization　on

lOO／oRe／HZSM－5　at　various　methane　specific　velocity

Fjg　2．　Benze．ne　formation　rate　with　time　on　stream　for　methane　aromatization　with

different　C　O2　concentration　on　10％Re／H：ZSM－5

Fig　3．　Conversion　and　p　roduct　formation　rate　with　time　on　stream　for　methane

aromattzation　with　2．　I　O／o　CO2　on　100／oRe／HZ　SM－5　at　1073　K　an　d　methane　SV　of

1　oooo　ml　／h／g－cat

Fig4．　Consumption　rate　and　conversion　ofCO2　and　formation　rate　of　CO　with　time　on

stream　for　methane　aromatizati　on　with　20／oCO2　on　l　OO／o　Re／HZSM－5

Fig　5・C・nversi・n　and　p・・duct　f・・’mati・n・at・舳tim，。n、t，eam　f。，　meth、n，

aromatization　with　20／o　C　O2　on　100／oRe／HZSM－5

Fig　6．　Conversion　of　CH4　and　CO2，　selectivity　of　higher　hydrocarbon　products　and

formation　rate　of　benzene　as　a　fun　ction　of　C　O2　concentration　in　methane　feed

Fig　7．　Conversion　of　CH4　and　CO2，　selectivity　of　higher　hydrocarbon　products　and

formation　rate　ofbenzene　as　a　function　ofreaction　temperature

Fig　8．　Conversion　of　CH4　and　CO2，　selectivity　of　higher　hydrocarbon　products　and

formation　rate　ofbenzene　as　a　fimction　ofmethane　specific　velocity
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F！’g　．1；．　Ben．zene　fp－rmatio．　n　rate　with　time　on　stream　for　methane　aromatization

at　different　specific　velocity　（SV）　of　methane　on　1096Re／HZS’M’T－5
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：ig　4，　Consumption　rate　and　conversion　of　CO2　and　CH4，　formation　rate　of　co

vith　time　on　stream　for　methane　aromatization　with　296　CO2　onlOOfoRe／HZSM－s
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Fig　5．　Conversion　and　product　formation　rate　with　time
on　stream　for　aromatization　of　methane　with　2％　CO
　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　2
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Fig　6．　Conversion　of　CH4　and　CO，　，　selectivity　of　HC　and　coke，

rate　of　benzene　as　a　function　of　COA　in　methane　feed　on　100fo
　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　　2
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rx

Fig　7．　Conversion　of　CH，　and　CO，，　HC　selectivity　and　formation

rate　of　benzene　as　a　function　of　teaction　temperature
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Fig．　8
р??@COnLv．ersion　of　CH4　and　CO，，　selectivity　of　hydrocarbon　products

and　formation　rate　of　benzene　as－＝@function　of　methane　SV
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Conclusion

1．　HZSM－5　supported　Mo　and　Re　catalysts　are　found　to　be

quite　active　and　selective　for　directly　converting　methane　to

benzene，　naphthalene　and　C2　hydrocarbons．　A　great

progress　for　methane　aromatization　has　been　made　because

of　the　discover　of　the　two　new　catalysts．

2．　Coke　deposition　on　the　catalyst　for　non一一〇xidative

conversion　ofmethane　has　been　solved　by　varying　methane

pressure　combined　with　addition　of　CO2　in　methane　feed．

Coke　selectivity　is　about　O　at　3　atm　methane　and　with　1－30／o

CO2　addition　in　metha　ne　feed．
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3．　Two　reactions　on　Re／HZSM－5　catalyst　are　identified：

i）　methane　with　CO　at　the　nearly　same　consumption　rate　to

produce　benzene，　naphthalene　and　C2　hydrocarbons；

il）　an　improved　reforming　reaction　of　methane　with　CO2　to

produce　aromatics　and　synthesis　gas．

4．　MoO3／HZSM－5　and　Re207／HZSM－5　are　frrst
endothermically　converted　to　its　reduced　state　before

productign　ofbenzene．　Mo　exists　in　its　carbide　and　is　highiy

dispersed　on　the　HLZSM－5　support．
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